SOME ASPECTS OF THE TONAL PHONOLOGY OF BODO

BY PRIYANKOO SARMAH

Supervisor Professor K. G. Vijayakrishnan Centre for Linguistics and Contemporary English

A DISSERTATION SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE AWARD OF MASTER OF PHILOSOPHY IN LINGUISTICS CENTRAL INSTITUTE OF ENGLISH AND FOREIGN LANGUAGES HYDERABAD 500007 JUNE 2004 © 2004 Priyankoo Sarmah

Acknowledgements

I express my gratitude to **Prof. K. G. Vijayakrishnan** for supervising this work. He helped me immensely in ranking the constraints correctly both in my academic and personal life during the two years of my M.Phil in CIEFL. In all senses he has been a real 'guru' for me!

Raji, for all the support I got during my M. Phil and especially for the unlimited supply of tea and coffee during my discussions with Vijay!

Ma, for bearing with my absence for the two years of my M. Phil and also for never complaining about it. Thanks for trusting me, always.

Deuta, for the constant support in all financial and academic matters. Thanks for taking interest in my work and for all the comments and inputs which helped in the shaping of my dissertation. (Thanks also for lending your car to meet my informants! You have been helping me ride smooth on rough roads!)

Boidehi, for being with me whenever I needed (and did not need!)! Whenever things slowed down, your persuasions made me work hard. You saw to it that even the tiniest thing in my life is set right!

During the two years of my M.Phil I was home only for 18 days and that is mostly because I had a home away from home on CIEFL campus! Thanks to **Prof. Amritavalli** and **Prof. Jayseelan**, they not only encouraged me (and many of us) to do better in my academic life but also never let me feel that I am 3000 KM away from home!

Prof. Madhavan, who has been a constant source of inspiration to me. I am always thankful to him for the help he extended, whenever needs arose!

Dr. M T Hany Babu, the best thing about you is that you are ready to answer even the stupidest questions we ask. Thanks for all the input I got from you to patch my boat to sail safe in the sea of linguistics!

Dr. Hariprasad, it was great fun working with you for the Assamese Syntax course! I wish you offered more courses like that!

Shrutidi (Dr. Shruti Sircar) has been playing the sisterly part throughout my stay at CIEFL. Your wit, systematicity and of course, your handwriting have been mesmerizing me for quite some time now!

There are some people on campus whom I can surely count on as my well wishers! Prof.

T Sriraman, Dr. Meera Srinivas, Dr. Anand Mahanand and Ms. Geetha Durairajan are some of the people who, I am sure, have been wishing the very best for me. Thanks a lot for the love and good wishes I have been receiving from you!

Prof. Robbins Burling, Professor Emeritus, University of Michigan, sent his paper on Bodo tones, as soon as I asked for it through e-mail! He also sent his valuable comments on many aspects of the findings of Chapter II and III of this work.

Dr. Caroline Wiltshire, Associate Professor, University of Florida, sent a lot of papers on tone languages, which played a vital part in the shaping of this work. I learnt a lot about field linguistics from her, which later helped me while gathering data for this work.

Dr. Tanmoy Bhattacharya, Lecturer, University of Delhi, for his encouragement and support. He made it sure that the best is available to us, whenever we visited Delhi University for conferences.

Dr. Phanindra Goyari, Department of Economics, University of Hyderabad. This work is not complete without the data and suggestions he provided me. I am always thankful to him for his interest in my work.

Purnima, the maidservant who got over all the shyness to speak into the microphone. Thanks for the data.

Mr. Gopinath Brahma and Mr. Robin Narzari, both teachers of Hem Barua Higher Secondary School, Ghoramari and Bodo poets too. The data collected from them constitute the major chunk of data used in this work. I thank them for their time, encouragement and comments.

Shakuntalaba (Shakuntala Mahanta), thanks for suggesting me to do an M. Phil in Linguistics and also suggesting the possible areas. Your hard work and academic excellence have always been an inspiration for me.

Ashtu, my classmate from the M.A. days who carries his village name with his given name and believes that the world is a big cricket ground with countries and continents mistakenly plotted over here and there! He has been sharing every happiness and sorrow of my CIEFL life. His intelligence and wit throws the bails off my wickets over and over again!

Jobin Matthew, another classmate of mine, whose sharpness and hard work often leaves me speechless! A person with a great sense of humour, I must say! Together we ate, drank and slept linguistics! Thanks for being there (and only **there**!), Jobin!

Mitr (Raju Tiwary), thanks a lot for everything you have been doing for me. I don't remember a single occasion when you said a 'no' to me!

iv

Parthada, thanks for accompanying me in my journey from my M.A. days till now. Your interdisciplinary interest is enough to make people like us run for cover! Thanks for being there in every high and low!

Bijoy Louis Dhanwar, research scholar, University of Hyderabad. Thanks Bijoy for doing everything possible and impossible to speed up my M. Phil work. Your friendship is a treasure for me!

Temsu, Research Scholar, CIEFL; Basan, Kamal, Chandan and Hindusthani: all students of University of Delhi, took interest in my work. I enjoy your company whenever we meet!

Alam bhai, thanks for always enquiring about the work done and to be done. Thanks also for ensuring that the research scholars' community does not starve during vacations. You are the best mess secretary!

Bibhutida, for all the valuable suggestions and comments on everything I did! The pleasure is all mine listening to you!

Rajatda, its people like you who have taught the basics of linguistics to many of us. Thanks for pushing me into the sea of linguistics!

Digantada and Kailashda for all your suggestions and encouragement. Digantada, thanks for patiently going through the first chapter.

Immanuel, whether you liked it or not, you produced the tones of Mizo whenever I wanted you to do so! Thanks for patiently listening to my discourses on tones. A section of my thesis is based on the data I got from you!

Chayanika, Elika, Gitashri, Kilang, Samir, Paroma, Pranesh, Ruth and Vineet: thanks for being such lovely juniors.

V

Angshu, Ravi and Pranjal, your presence in the hostel is enough to keep the hostel alive!

Pathak (Mahananda), thanks for never letting the laughter fade out in the hostel. Thanks also for helping me solve the puzzle of the missing link in the evolution of man from monkeys!

Rahul (nee Tinku!), thanks for being readily available for any help. I admire your organizational capabilities!

Rahul, thanks for all the help I got from you!

Pandeyda (**Mr. K.N. Pandey**), thanks for all the encouragement and for seeing to it that we do not suffer from malnutrition on CIEFL campus! Thanks for the large variety of fruits you have been feeding us all these years!

Li (Alipokla Longkumer), for patiently listening to me over endless cups of coffee! It's a pleasure talking to you!

Jobin Thomas, for all the technical help! Your CD writer is a real blessing for all of us. Thanks for letting us use it and never complaining about its excessive use!

Anuradha, thanks for being such a lovely friend and for all the gastronomical sojourns you took me to!

Chetan Pinto for helping in taking the printouts of the thesis! If one smells Udupi in the thesis, it's him!

People in the **Dean's Office**, thanks for making all the rules look so simple! I enjoyed your pulling my leg during my stay in CIEFL!

Library staff, for helping in keeping the library in a much better shape than many of the higher education institutions in India. The best thing about is that you always acknowledge our ideas and suggestions.

Mess workers, thanks for the love and good wishes. A wish coming from your end makes the day for many of us!

Raju, Balram, Ishwar and Madan for the help in violating copyrights! Long live Sai Xerox!

Hostel Office, for the support and timely help they have been providing me.

The Cafeteria, where most of the linguistics in CIEFL happen!

Abstract

This dissertation is an attempt to answer the basic questions regarding the tonal phonology of Bodo.

Chapter I gives a general introduction to the Bodo language, the existing research work in Bodo and summarizes the framework followed in the present work. This chapter also gives a brief explanation of the term 'Tone Language' and tries to exemplify how tonal contrasts occur in tone languages.

Since the literature on Bodo (specifically tones) is confusing as contradictory claims are made, we planned to establish the phonetics and phonology of Bodo tones with the help of digitally recorded data.

Chapter 2 explains how tone assignment takes place in Bodo non-derived monosyllabic lexical entries. This chapter also looks into the segmental interferences in realization of pitch. It also gives an overview of the method followed for data collection and analysis in this research work.

Chapter 3 looks at the tone assignment in Bodo non-derived disyllabic lexical entries. Like the previous chapter this chapter also looks into the segmental effects in the realization of pitch in disyllables.

Chapter 4 examines the tone assignment pattern in derived polysyllabic entries. It looks at the alterations that suffixes go through in a derivation. It is also an attempt to identify and explain different morphophonemic patterns observed in Bodo.

Chapter 5 gives an overview of the OT (Optimality Theory), and an OT account of the tonal phenomenon including tonal alteration in Bodo.

Table of contents

Acknowledgements	ii
Abstract	V
Table of contents	vi
Chapter I : Introduction	
1.1 Introduction	1
1.1.1 The Bodo Language	1
1.1.2 Phonemes in Bodo	2
1.1.2.1 Consonants	2
1.1.2.2 Vowels	2
1.1.3 Tone languages	3
1.1.4 Tones and the Bodo language	4
1.1.5 Studies on Bodo tones	4
1.2 The present study	5
1.3 Theoretical Background	6
1.3.1 Optimality Theory	6
1.3.2 The Autosegmental nature of Tones and the Optimality Theory	7
1.4 The plan of the dissertation	9

Chapter II : Tonal Patterns in Monosyllables

2.1 Introduction	11
2.2 The Data and The Method	14
2.3 The Hypotheses	15
2.4 The Findings	18
2.4.1 The Onset Effect	19
2.4.2 The Tones	20
2.4.3 The Nasal Coda	21
2.4.4 Vowel Length in Contrastive Tones	23
2.4.5 The Glottal Stop Phenomenon	24

2.4.6 The $/r/$ in the Coda	25
2.5 Conclusion	26

Chapter III : Tonal Pattern in Disyllables

3.1 Introduction	30
3.2 The Hypothesis	30
3.3 The Findings	31
3.3.1 Tone Assignment in Disyllables	31
3.3.2 The Onset Effect on the First Syllable	33
3.3.3 Tone Distribution and Onset Effect on the Second Syllable	35
3.3.4. '0' Tone in Disyllables	38
3.3.5 The Tones	41
3.4 Conclusion	41

Chapter IV : Some Issues in Bodo Morphophonemics

4.1 Introduction	43
4.2 The Hypothesis	43
4.3 The Findings	46
4.3.1 Plural Clitics Manoy, Burza and Gobang	46
4.3.2 Alteration In Ph-, Bv-, Sv Type Of Causative Prefixes	47
4.3.3 –sör and –phör type suffixes	49
4.4 The Exceptions	52
4.4.1 The causative suffix-ho	52
4.4.2 –mon type alteration	55
4.5 Conclusion	56

Chapter V : An OT Analysis of Bodo Tones	
5.1 Introduction to Optimality Theory	59
5.2 OT Treatment of Tones	61
5.3 OT Treatment of Bodo Tones	66
5.3.1 Introduction	66
5.3.2 A non-derived lexical item must be specified with a tone	67
5.3.3 A TBU without a lexical tone is specified with a default mid tone	71
5.3.4 Tone assignment in Bodo derived Polysyllables	81
5.3.4.1 The general pattern	81
5.3.4.1.1 Type I (a) derivation	82
5.3.4.1.2 Type I (b) derivation	84
5.3.4.1.3 Type I (c) derivation	86
5.3.4.2 The –mōn suffix	87
5.3.4.3 The <i>-ho</i> suffix	92
5.4 Conclusion	95
Appendix I	96
Appendix II	100
Appendix III	207
Bibliography	208

Chapter I Introduction

1.1 Introduction

1.1.1 The Bodo Language

Bodo is one of the major Tibeto-Burman tone languages, spoken in many parts of the North-Eastern states of India as well as in parts of West Bengal and Nepal. In Assam, Bodo is predominantly spoken in the districts of Goalpara, Nalbari, Barpeta, Dhuburi, Kamrup, Kokrajhar, Darrang, Sonitpur, Lakhimpur, Nagaon, Marigaon, North Cachar and Karbi Anglong. The fact that it is now included in the eighth schedule of the Indian constitution, demonstrates its socio-political importance. According to recent Indian government census Bodo is spoken by almost a million speakers. But these figures are highly unreliable as even during this research we found many people, who report to the census authorities that they speak Bodo, do not actually speak or in worst cases know the language. They just use the term "Bodo" to secure their ethnic identity.

As reported by Bhattacharya (1977) Bodo has at least four varieties:

- North Goalpara variety, spoken in the northern regions of Goalpara and Kamrup districts. However the North Goalpara and the North Kamrup variety have salient differences.
- 2. A variety spoken in South Goalpara, Garo Hills and South Kamrup.
- A variety spoken in Darrang, Lakhimpur and a few places of Arunachal Pradesh, Called the north central Assam dialect.
- 4. Another variety spoken in the Nowgong, North Cachar and Karbi Hills.

During this research we saw possibilities of assimilation between these varieties. Hence it

is pertinent to conduct a research to identify the varieties of Bodo anew.

1.1.2 Phonemes in Bodo

There are 16 consonants and 6 vowels in Bodo. One striking feature of Bodo is the existence of a rather large phoneme inventory, which is rare in the North-Eastern Tibeto-Burman tone languages. The consonants and vowels of Bodo are described as below:

1.1.2.1 Consonants

Writing convention in this thesis	Description (Bhattacharya, 1977)
ph	Voiceless bilabial aspirated plosive
b	Voiced bilabial plosive
th	Voiceless denti-alveolar aspirated plosive
d	Voiced denti-alveolar plosive
kh	Voiceless aspirated velar plosive
g	Voiced velar plosive
m	Voiced bilabial nasal
n	Voiced denti-alveolar nasal
ng	Voiced velar nasal
S	Voiceless alveolo-palatal fricative
Ζ	Voiced alveolo-palatal fricative
r	Voiced denti-alveolar tremulant
1	Voiced denti-alveolar lateral
h	Frictional semivowel
У	Frictionless palatal semivowel
W	Frictionless rounded velar semivowel
Table 1.1	

1.1.2.2 Vowels

Writing convention in this thesis	Description (Bhattacharya, 1977)
i	Front, high, unrounded
u	Back, high, unrounded
e	Front, mid, unrounded
ō	Central, mid, rounded
0	Back, mid, rounded
a	Central, low, unrounded

Table 1.2

1.1.3 Tone languages

If the pitch of a word can change the meaning of a word, that language is called a tone language. The pitch not only changes the nuances of the language but also changes the core meaning of the language. It is worth mentioning that according to some studies 60-70 percent of world's languages are tone languages!

Yip (2002) mentions Hyman (in press) as a plausible definition of a tone language:

'A language with tone is one in which an indication of pitch enters into the lexical realization of at least some morphemes'

To see how pitch change results in differences of meaning we can look at an example from Cantonese, where the syllable [yau] can be produced in six different pitches, which has six different meanings as shown below:

[yau]

high level	'worry'
high rising	'paint (noun)'
mid level	'thin'
low level	'again'
very low level	'oil'
low rising	'have'

In some languages, tonal distinction in polysyllabic words is obtained by contrastive positioning of restricted tones in different syllables. For example in Dagaare, a Gur language spoken in Ghana, a disyllabic word can be specified with two tones, H(igh) and L(ow) in the following ways:

In case of a disyllabic entry [yuori], tonal distinction is noticed in the following way:

LH [yuori]	'penis'
HL [yuori]	'name'

In many other languages, the positioning of the lexical tone does not matter much. It may appear anywhere in the lexical entry. The exact location of the tone may change according to the morphological or phonological environment.

1.1.4 Tones and the Bodo language

Like many other Tibeto-Burman languages, Bodo is also a tone language. However dearth of proper research of its tonal phenomenon has left many questions about the language unanswered. There have been conflicting views about the number of tones (ranging from zero to four!), tone assignment system and morpho-phonemics of the language. The only agreement most of the researchers have about the language is that it is a tone language. This thesis tries to settle many issues regarding the tonal phonology of Bodo definitely.

1.1.5 Studies on Bodo tones

In spite of being spoken by almost a million speakers, Bodo is not a well-researched language. The articles and books I had access to are Halvorsrud (1959), Burling (1959) Bhattacharya (1977), Weidert (1987), Boro (1991) and Joseph and Burling (2001). However, it is very disturbing that to come across contradictory claims regarding Bodo tonology in the research cited above. Our initial task was to go through the literature and establish the facts to our satisfaction. Since we consider the details of previous research as a point of departure for our own in the following chapters, we do not intend to give a detailed summary of the previous findings at this point.

1.2 The present study

Since Bhattacharya (1977) claimed the maximum number of tones in Bodo, we thought that his data would be the best starting point for our work. We gathered 128 words with distinct tonal features according to Bhattacharya (1977) and crosschecked them with the native speakers to further refine the set. This set constituted our database. We made two speakers of Goalparia dialect, use the words in the sentence frame (2) and digitally recorded the utterances.

(2) Anjalua_____ bungdungmon

Anjalu _____ said

Initially we had the idea to substitute the proper name *Anjalu* with the first person singular pronoun /aŋ/. But later we realized that [ŋ] has a tendency to raise the tone of ethe following sequence. Therefore we chose a proper name so that the nominative marker /a/ would be word final. We give further details of our methodology in Chapter II.

1.3 Theoretical Background

1.3.1 Optimality Theory

Though the II, III and IV chapter of this work are mostly descriptive, the findings regarding the tonal pattern of Bodo is analysed in the framework of Optimality Theory (Henceforth OT) in Chapter V. OT is a linguistic theory proposed by Prince and Smolensky (1993) McCarthy and Prince (1993 a, b). It is an extension of the Generative Grammar. OT assumes that the surface forms or the outputs are the results of conflicts between different constraints. A constraint in OT is a rule that applies pressure to a phonological system. The constraints may induce pressure to change things (markedness constraints) or to preserve things (faithfulness constraints). In principle, constraints are violable in OT. These constraints are present in every language but they are ranked differently. Language specific ranking of the constraints explain the differences between languages. The core issues of OT are discussed in Chapter 5 of this work.

Though OT is an extension of the UG, its assumptions about UG are radically different from the rule based derivational theories. OT addresses several problematic issues. Archangeli and Langedon (1997) summarise some of the issues cited below:

- It defines a clear and limited role for constraints.
- It eliminates the rule component entirely.
- It focuses research directly on language universals
- It resolves the "non-universality of universals" problem.

1.3.2 The Autosegmental nature of Tones and the Optimality Theory

Goldsmith (1976) dealt with the problems of expressing tones suprasegmentally. Autosegmental representation of tones takes the form of a segmental representation of the vocalic and consonantal matrices linked by association lines to tonal features. In the autosegmental model, tone is represented separately from the segmental base of an utterance but the tonal features are associated with segments. Goldsmith establishes the general conditions of the wellformedness:

1. All vowels are associated with at least one tone.

- 2. All tones are associated with at least one vowel.
- 3. Association lines do not cross.

The first two constrains of the wellformedness condition see to it that there are no stranded unpronounceable tones or vowels. However, it does not restrict the number of ways in which tones and Tone Bearing Units (TBU) are associated. In order to limit these possibilities some scholars take up the principle that tones and TBUs should be associated from left to right (Goldsmith, (1976), Williams (1976), Haraguchi (1977), Clements and Fort (1979), Goldsmith and Clements (1984)).

This principle came to be known as the Universal Association Convention. Later scholars noticed that though UAC is appropriate in many languages, there are cases where the final TBU is associated with several tones but the default situation is not to have such multiple associations. The convention can be reformulated as in (1):

(1) Map a sequence of tones on to a sequence of tone bearing units,

(a) from left to right

(b) in a one-to-one relation.

Condition (1)b precludes multiple associations of tones to tone bearing units. The convention is only the default case. It may be overridden if so stipulated for specific languages.

However, condition (1)a is not always appropriate. There are cases in which it s not possible to associate them from any direction. In these cases the tones are lexically determined and hence, not predictable. Only the remainder will be assigned by rule. Goldsmith brings in a particular formal device for the initial lexically determined association, i.e. the star notation. By marking the particular tone and the TBU to be associated, with a '*', and by associating these first, the desired alignment is achieved. This device identifies a particular TBU as the center of the tone pattern.

It is assumed that identical tones in sequence constitute a single occurrence of a tone on the tonal tier: sequences of similar tones on this tier are not permitted. This is known as the Obligatory Contour Principle (OCP). The principle was put forward by Leben (1973a) and then formulated by Goldsmith (1976:36) as follows:

OCP

At the melodic level of the grammar, any two adjacent tonemes must be distinct. Thus HHL is not a possible melodic pattern: it automatically simplifies to HL.

The implication of the model for the place of tone in prosodic structure is more important than the use as a descriptive tool. Yip (2002) proposes a general theory of tonology in the OT framework, which can account for all the autosegmental representations. This set of constraints is discussed in Chapter 5 of this work. The inherent strength of OT in dealing with typology makes it the most favourable framework to account for a system as complex and varied as tone.

1.4 The plan of the dissertation

This dissertation is an attempt to definitively answer the questions regarding tonal phonology of Bodo. This dissertation is in five chapters.

Chapter 2 explains how tone assignment takes place in Bodo non-derived monosyllabic lexical entries. This chapter also looks into the segmental interferences in realization of pitch. It also gives an overview of the method followed for data collection and analysis in this research work.

Chapter 3 looks at the tone assignment in Bodo non-derived disyllabic lexical entries. Like the previous chapter this chapter also looks into the segmental effects in realization of pitch in disyllables. Chapter 4 finds out the tone assignment pattern in derived polysyllabic entries. It looks at the alterations that suffixes go through in a derivation. It is also an attempt to identify different morphophonemic patterns observed in Bodo.

Chapter 5 gives an overview of the OT (Optimality Theory), and an OT account of the tonal phenomenon in Bodo.

Chapter II

Tonal Patterns in Monosyllables

2.1 Introduction

While talking about the tonal properties of Bodo, Weidert (1987) is of the opinion that the tone patterns in Bodo are dependent on the syllable types and the consonantal specification of the syllable coda. According to him the 'smooth' and the 'stopped' syllable types give rise to different pitch patterns on the following syllables. He presents the smooth syllables as of /CV/ and /CVR/ types where R represents the resonants /m, n, n, r, u, i/. The stopped syllables are of /CV?/, /CVR?/, /CVP/ types where P represents the obstruents /p, t, r?, k/. According to him monosyllables do not show any pitch difference but the voice level differs in terms of broad mid or normal (it is not quite clear what exactly he means by that). However, in the case of connected speech, the pitch of the word in the second position depends on the pitch of the first word, which is utterly predictable from its coda consonant. If the first syllable has an underlying glottal stop, the glotallization is lost but it gives rise to a high pitch on the following syllable. But high pitch is NOT noticed on the second syllable if the first syllable is a smooth one. The main generalizations regarding tones in Bodo, according to Weidert, are

- 1) tone contrasts are not present in monosyllables and
- tones are to be derived from syllables codas which may or may not be realized on the surface.

The implication is that Bodo is not a full-fledged tone language but a case of tonogenesis. We show, in this chapter, that both these generalizations are untenable in the

11

contemporary dialect of Bodo that we analyzed (Goalpara dialect). However, Weidert himself gives examples where the predictable pitch phenomena show exceptions. There are instances of a first syllable having a glottal stop, but failing to produce a high pitch on the second syllable. He marks the compounds with /doi? / as irregular. He says, "Such an interpretation produces the (structurally unwanted) effect of underlying allomorphy under unpredictable conditions and thus tends to blur the morphological-cum-tonogenetic unity of the system" (Weidert, 1987: 422)

As we said earlier, we will illustrate with spectrographic evidence our claim pertaining to contrastive tones in Bodo. We will demonstrate that monosyllables with the same C and V specification can have different, underlying, lexical tones, which are not attributable to any consonantal margin of the syllable. However, when we examine polysyllables in Chapter IV, we will show that Weidert's claim that the tonal specification of the preceding syllable is realized on the following syllable is partially correct in certain types of morphologically derived forms.

Baro (1991) says that Bodo has a two-tone system, which he describes as the rising and the falling tones. The syllable with a falling tone, according to him, has a lengthened vowel without any glottal check. The glottal stop, according to him, occurs only in syllables with a rising tone with one of the consonants /r, m, n, ŋ/ as a coda. We can infer from Baro's observation that, he too, like Weidert, associates a glottal stop or a glottalic consonant with high tones. His conclusion is that Bodo has unpredictable i.e., lexical tone and that every lexical entry must be specified for tone as well.

We will show in the course of this chapter and the next, that as claimed by Baro, Bodo does have lexical tone even on monosyllables and that only high tone is associated with

12

the glottal phenomenon. However, when we consider disyllables, in Chapter III, we will show that not every syllable of a lexical entry may be specified for tones.

Bhattacharya (1977) claims that Bodo has a four-tone system, with three tones available in different positions and a neutral tone, which is dependant on either the preceding or the following non-neutral, toned syllable (Bhattacharya, 1977). According to Bhattacharya the three tones in Bodo are high, mid and low (indicated by /1/, /2/ and /3/ respectively). He identifies tone1 as having a level or a rising pitch pattern. Tone2 as a level or a falling and tone 3 as having a falling pitch pattern.

Halvorsrud (1959) also identifies three tones, in his early account of the Bodo language, as high, medium and low. He observes that the difference between the high and the low tone is very evident. But he does not talk much about the mid tone, as it lacks perceptual salience. In the absence of any data exemplifying the "third" tone we must come to the conclusion that Bodo has only two lexical tones according to Halvorsrud.

Joseph and Burling (2001) come to the same conclusion when they say that Bodo has only a two-tone system. Earlier in Burling (1959), evidence for only two tones was found, i.e. high and low. He even postulates that the high tone ends with a glottal stop but at the same time he speculates the possibility of a high tone not associated with a glottal stop.

Our findings based on extensive data lead us to concur with the findings in Burling (1959) as well as Joseph and Burling (2001). There are only two tones in Bodo and the glottal stop is only idiosyncratically associated with the high tone in the synchronic variety of Bodo.

2.2 The Data and The Method

The analysis in this thesis is based on the data collected from two male, educated, adult native speakers of Bodo (the Goalparia dialect). The data of 59 sets of words were constructed using Bhattacharya (1977) as the resource work, assuming a three-way tonal contrast. The native speakers were then asked to pronounce the words placing them in the blank position of the sentence frame- "Anjalu¹ _____ said"². The data was recorded on a portable Digital Audio Tape (DAT) recorder, which ensures recording with minimum distortion.

The words that were selected had contrastive tones, as claimed by Bhattacharya (1977). For example a set of words like the following was used for the experiment (see Appendix

1):

Set 1

¹bar *v*. jump with a high tone
²bar *n*. wind with a mid tone
³bar v.bloom with a low tone.

Set 2

¹dōy	v. lay eggs, place; on a head	with a high tone
²dōy	<i>n.</i> water	with a mid tone
³ dōy	v. sweeten, feel sweet	with a low tone.

Set 3

¹ ray	v. harm by magical power	with a high tone
² ray	v. abuse	with a mid tone
³ ray	v. speak	with a low tone.

(Bhattacharya, 1977)

¹ A proper name.

² The sentence in Bodo is "Anjalua_____ bungdungmon."

The informants were given a list of Bodo monosyllables in Roman script and also in Devanagari, which is the official script for Bodo now, along with the gloss to disambiguate homophones. They were required to insert the given word in the sentence frame given above. The data were recorded with a digital recorder for optimum clarity and were analyzed by speech analysis software called PRAAT 4.0.26. Each word was pronounced four times and out of the four iterations three were selected for analysis. The data were then fed into the computer with an analog interface cable. With the help of PRAAT 4.0.26 the words were segmented from the sentence. With the help of the different graphic representations created by the software, the pitch and the sound quality were determined.

2.3 The Hypotheses

In this study we find support for two hypotheses:

1. Though it was claimed that Bodo has four tones (Bhattacharya, 1977), there are actually only two tones available in Bodo monosyllables. We will look into the characteristics of the tones as we put forth our arguments for our hypothesis. Our hypothesis is contrary to Weidert, who claims that Bodo tones are predictable and that tonal distinctions do not show up in monosyllabic words.



15



²lao 'gourd'



Contrary to the claims of Weidert, the pitch tracks noticed in the pictures above clearly show that an unpredictable tonal difference exists in Bodo monosyllables.

¹hor 'send'

²hor 'night'











Fig 2.7

However, a look at the pictures above clearly shows that *hor* does not have a three-way distinction of tones as postulated by Bhattacharya. We observe only a two-tone system in Bodo even in our analyses of the other monosyllables. The system we postulate has only a high tone and a low tone.

2. There may be a glottal stop at the end of an open syllable carrying a high tone. If the syllable is closed then the consonant may be glottalized. It challenges Weidert (1987), who proposes that some of the Bodo monosyllables acquire the high tone due to a glottal stop that occurs in case of open syllables. We will try to show that the tone is not affected by the glottal phenomenon but the glottal phenomenon is associated only with the high-toned syllables.



³za 'be, become'



¹gong 'horn'

³gong 'bent, bend'



The pictures above clearly show that the glottal stop may occur with an open or a closed syllable, but only in the context of high tones.

2.4 The Findings

While analyzing the data it was noticed that monosyllables follow a particular pattern as far as the tone distinction is concerned. In monosyllables, the pitch pattern (if any) on the onset was ignored, and the pitch pattern on the nucleus and the coda (if any) was considered to be the tone pattern of the word. The pitch, which showed a frequency of 100-110 Hz (for both the male speakers), was considered to be of a *medium* tone (which we call 'm') and the any pitch pattern above and below it is considered to be of a *high* (+H) and *low* (L) tone respectively.







In the pictures above we see the two-tone patterns we describe to be high and low. Figure 2.14 shows an instance of a high tone, whereas figure 2.15 shows a low tone associated with the word *ha*. We observe that both the pitch initiations are almost at the same position. We associate this position with the mid tone. Hence we come to the conclusion that all the tones start from a default tone, which we call in our analysis the mid tone from thereon it either rises or falls to assign a distinct tonal identity to the syllable.

2.4.1 The Onset Effect

The analysis also showed that the initial pitch level on the analyzed part of the word differed depending on the consonant in the onset position. In case of an aspirated onset the pitch started at a high position and if it is associated with a high tone it moved to a further high position. Whereas, the onsets having other consonants, i.e. voiced obstruents and sonorants started at a slightly lower mid position and moved to a relatively higher position for the lexically specified higher and to a lower pitch for the other tone.



¹gab 'colour'



³phung 'be fat and fleshy'

³gab 'weep'



It can be concluded from this observation that the pitch initiation is affected by the consonant quality of the onset that is extremely predictable. This predictable raising/lowering effect is ignored in the rest of the dissertation.

2.4.2 The Tones

It is observed that the pitch pattern rises from mid to high in Tone 1 as seen in the picture below. Tone 1 may be accompanied by a glottal stop or a glottalized consonant in the coda position.



To reiterate the point we have made earlier, we could not find out any systematic difference between tones 2 and 3 of Bhattacharya. A look at the pictures below makes this claim evident.



A detail look at the CVG pattern of monosyllables (where G is a glide) clearly shows a two-way distinction between the tones. The low fall observed in some of the words having supposedly Tone 2 and Tone 3 systematically failed to show any difference (as in the pictures above).

2.4.3 The Nasal Coda

In our analysis we observed that whenever a nasal consonant occupies the coda position, it takes the rising pitch when the word is associated with a high tone. In the words with a low tone, the pitch on the nasal coda shows a tendency to rise towards the end of the consonant. The pictures below show that the nasal coda continues to rise in the context of

³beng 'agree'





¹ham 'be lean and thin'







²khon 'sing, occasion'



a high tone and even in the context of a low tone it induces a slight rise towards the end. We therefore assume that whereas nasal codas have the property of raising the tone, nasal onsets do not have that property as they induce a lowering of the initial 'mid' tone on the vowel.

2.4.4 Vowel Length in Contrastive Tones

We observe a particular pattern of vowel duration in open syllables. The pictures below make it evident that the vowel length is affected by the tonal specification. It is

¹**bu** 'strike, beat'

³bu 'swell'



seen from the pictures that the vowel in the word with a low tone is distinctively longer than the vowel in the word with a high tone. This is the case uniformly present in all the open syllables as Fig. 2.31 and Fig. 2.32 illustrate:



²ha 'be able'



2.4.5 The Glottal Stop Phenomenon

In case of Tone 1 we sometimes see a glottal stop or a glottalized consonant at the end of the syllable. Whereas, low toned syllables do not ever show a syllable final glottal stop. Spectrographic analysis of monosyllables with high tone shows random presence of a glottal stop. We call it random presence as it fails to show up regularly even in the same lexical entry. In Fig. 30 and 31 we see two different iterations of the same lexical entry ¹gong 'horn'



¹gong 'horn'

by the same speaker differs in terms of the presence of a glottal stop³.

The glottal stop induces a sharp fall in pitch as it does in Mizo (Lalrindiki, 1989). But the basic difference between the occurrence of the glottal stop in Mizo and Bodo is that in the former it occurs in a low tone syllable, where it makes the low tone an extra low tone actually realized as a very sharp fall. Whereas in case of the latter the glottal stop occurs in high tone syllables. Among the set of data collected the following words have a glottal stop as a coda in a high tone syllable:

/hu/	'land'	/kha/	'pluck'
/khu/	'make naked	/ro/	'leak out'

³ It is clear that a nasal coda has a tendency to raise the tone and the glottal stop '?' to depress the tone. The presence of a ? actually pushes the H tone to the left (as seen in Fig 2.30) and if the nasal is in the final position, the H tone is free to move to the right. Of course, this is just a tentative observation. More work needs to be done in this area.
/si/ 'cloth' /song/ 'cook' /sa/ 'north/make rope' /su/ 'wash/saw/thorn'

/za/ 'eat'

It seems to us that some high tones lexical entries have an idiosyncratic ? underlyingly. The deletion of the glottal stop (which seems to be in free variation) is optional depending on factors yet to be established.

2.4.6 The /r/ in the Coda

In our analysis of the Bodo data we found some abnormalities in the pitch associated with the consonant r/r in the coda position. It was noticed that in some cases the pitch track

¹er 'move by finger or handle'

³er 'grow, increase'



³bar 'bloom'

³bar 'bloom'



associated with the /r/ is rising and in a few cases it is actually falling. We notice that the pitch on 'r' rises in an unsteady manner in both high and low toned syllables; e.g. Fig 2.35, 2.36, 2.37 and 2.38. However, to our dismay we found the pitch falling, in a tone syllable with 'r' in one of the utterances of *bar^L* '*wind/bloom*'. We then took a look at all the instances of 'r' in the coda in our recording in both high and low toned syllables. The irregular rise in tone was observed in 60% of the iterations. However the pattern remained consistent (unlike the glottal stops) in specific items. We are open to the idea that there could be two 'r' s in Bodo with distinct phonetic patterns of realization. We leave the issue for future research.

2.5 Conclusion

We conclude that in Bodo the tonal phenomenon is lexical. In case of monosyllables we see that they have a tonal distinction, however it is not a three-way distinction, as claimed by Bhattacharya, but only a two-way distinction.

It is clear that the right edge of the rhyme is associated with the lexically specified tone (+H or - H) and the default pitch on the left of the lexical tone is a mid tone which we depict as 'M'.

It could be argued that the lack of distinction between Bhattacharya's Tone 2 and Tone 3 is actually a case of mistaken homophony, i.e. the lexical item with Tone 2 is taken to be a lexical item with Tone 3 and vice-versa. Our answer to this objection is that this interpretation does not

explain why, then, the informants did not systematically mistake lexical items with Tone 1 to be homophones with Tone 2 or Tone 3. Moreover, as we said earlier, our set of words for determining the tonal distinction is fairly large, i.e. 59 set. Hence we can safely say that our findings are quite robust.

Having found no evidence for a three way tonal distinction, but sufficient evidence for consonantal effects on the pitch pattern (see section 2.4.1), we then selected words with the pattern VG in the rhyme to avoid any consonantal effect in the tone. We avoided selecting monosyllables with a short vowel to weed out checked syllables at the initial pass. Then we examined 9 sets of monosyllables with the VG rhyme sequence to see whether our hypothesis regarding a two-tone system was robust or not. We are happy to conclude that of the 9 sets of the tokens examined we had 100% success in isolating the two types of tonal pattern in Bodo. The pictures below show the pitch pattern with the VG rhyme sequence:

gao^{+H} 'tear or split' **gao^L** 'shoot by arrow or gun'



Fig 2.39

law ^L 'gourd'

law^{+H} 'be long'





Figure 2.39 and 2.41 clearly show a rise in the pitch as it represents the +high tone. In figure 2.40 and 2.42 we can notice a fall that can be associated with a –high or a low tone.

We have shown that the glottal stop is associated only with high-toned syllables. However we have also shown that not all high-toned syllables end in a glottal stop. Whereas some lexical items do not have a final glottal stop in any of the iterations, some lexical items exhibit a final glottal stop at least in some of the iterations (Figure 2.33). There are two explanations to this phenomenon.

(1) One can postulate a glottal stop underlyingly with all overtly high-toned syllables (on the lines of Weidert). Then an optional rule of glottal stop deletion is postulated to account for the absence of glottal stops in some words. This explanation makes Bodo out to be a non-tone language underlyingly requiring a rule of +H tone insertion in the context of a glottal stop, supporting Weidert. But the problem with this explanation is that even for lexical items that do not have a final glottal stop overtly in any of the iterations, we end up postulating an abstract, final glottal stop. The second problem with this solution is that an abstract glottal stop requires to be postulated only at the right edge of some lexical items. If the glottal stop is a legitimate segment in synchronic Bodo, why is it confined to the morpheme final position?

(2) The second explanation, to which we are sympathetic, is to assume that Bodo is a protone language with every lexical item associated with one of the two tones. Further, some high tonal syllables may have an underlying glottal stop that may or may not be realized overtly. It requires an optional rule of glottal deletion. In answer to the criticism that only +H syllables should be associated with a glottal stop, we suggest that only a detailed investigation of the effects of tones on segments and their allophones will yield an answer to this question.

It will be recalled that high-toned syllables induce a shortening of the vowel. This could be one of the realizations of a 'tense' articulation associated with the high tone (Bhattacharya, 1977). However, more research is required before we can definitely claim that high tones induce other specific segmental effects in Bodo.

Chapter III

Tonal Pattern in Disyllables

3.1 Introduction

Some facts that are noticed in Chapter II can be reiterated in the analysis of disyllables. For example, as in monosyllables we notice no difference between Tone 2 and Tone 3 in case of the disyllables. Contrary to Bhattacharya (1977), we will try to show in this chapter that even in disyllables Tone 2 and Tone 3 are not distinct.

Weidert (1987) claims that the tonal specification of the preceding syllable is realized on the following syllable. In relation to his claims we will show in this chapter that the preceding syllable of a disyllabic word, in fact, does not take any tone at all. It is the final syllable where the tone is realized.

3.2 The Hypothesis

The hypotheses in this chapter regarding disyllables will be:

- 1. There are only two tones in Bodo disyllables the high and the low tone.
- 2. Disyllables cannot take tones in both the syllables.

* T_iT_j (when *i* and *j* are distinct)

 $T_i T_i$ (when both the syllables take identical tonal pattern)

3. Only the final syllable is specified for tone.

3.3 The Findings

3.3.1 Tone Assignment in Disyllables

In the tone assignment of disyllables we observe an interesting system. Contrary to Bhattacharya we see that the pitch of the first syllable of a disyllabic word always remains between 100-110 Hz, this according to us is the mid tone. As this system of tone assignment is uniformly followed, it is pertinent to say that in Bodo the first syllable of a disyllabic non-derived word does not take any lexical tone. The mid tone, which is assigned to the first syllable of the word, is actually the default or the unmarked tone. We present below spectrographic evidence that we found while analyzing the data. However, the transcription of the words from 3.1 to 3.4 is as found in Bhattacharya (1977).



²bi³dōy [^Lbi^Ldōy] 'egg'



²da¹lang [^Lda^{+H}lang] 'do not take'



³da³lang [^Lda^Llang] 'bridge'



Fig. 3.4

Consider the pictures in fig. 3.1 and fig. 3.2, where the first syllable in ${}^{+H}bi^{+H}d\bar{o}y$ (fig.3.1) is supposed to have a high tone and the first syllable in $[{}^{L}da^{+H}lang]$ (fig. 3.2) is supposed to have a 'mid' tone according to Bhattacharya. What we, in fact, see is our mid tone in both the initial syllables. Had the first syllable been associated with a high tone, the pitch track would have been like as it is shown in fig. 3.5, which, of course, is manipulated for the purpose:

¹bi¹dōy [^{+H}bi^{+H}dōy] 'juice'



Fig. 3.5

Similarly, take the case ${}^{L}bi^{L}d\bar{o}y$ (fig. 3.3) and ${}^{L}da^{L}lang$ (fig. 3.4). Had the tonal specification on the first syllable been low, the pitch pictures would have looked like fig. 3.6 and fig. 3.7 respectively.



³da³lang [^Lda^Llang] 'bridge'



The fall should have been initiated in the first syllable itself and the pitch should have kept falling as in the 'corrected' pictures show. But this is not what happens in the language (fig. 3.3 and fig.3.4).

Looking at 3.1, 3.2, 3.3 and 3.4 we can say that only the final syllable is associated with a lexically specified tone. The first syllable in all these cases is the default mid tone. Therefore our claim is that in Bodo, each word must have one and only tonal specification which is associated with the syllable on the right. The preceding syllables get a default mid tone. Therefore the transcription of the words is modified hereafter. Henceforth, we follow the practice of indicating lexical tone only on the final syllable.

3.3.2 The Onset Effect on the First Syllable

In Chapter II we observed that the pitch of the syllable is influenced by the consonant in the onset position. Similarly in case of the disyllables the default tone of the first syllable is influenced by the consonant in the onset position. It accounts for the pitch variation in the mid tone from 100 Hz to 110 Hz. As observed in Chapter II, the pitch of a TBU is raised by an aspirated onset. As seen in the picture below the aspirated consonant in the onset position raises the pitch. The fall seen in the pitch can be interpreted as a process of lowering the pitch to the unmarked position.

kha^Llay 'do, perform'



Similarly we notice in case of the sonorants /s/ and /h/, that the initial pitch is raised by these consonants in the onset position. But the consonants /l/ and /m/ in the onset do not induce a high tone. They carry the default mid tone.



mo^Ldom 'sweet smelling'







Fig 3.10







Fig 3.11

Fig 3.12

Figure 3.9 and 3.10 show that the onset does not induce a high or a low tone, rather it associates itself with the default 'mid' tone. Hence they do not induce a fall in the mid tone. Whereas figure 3.11 and 3.12 show a drop in the pitch, which indicates that the consonants /h/ and /s/ in the onset induce a high pitch.

The tone of the second syllable of a disyllabic word is constructed in comparison with the default tone in the first syllable. So any pitch pattern higher than the default tone can be interpreted as the high tone, conversely any tone lower than that of the default tone can be seen as a low tone.

3.3.3 Tone Distribution and Onset Effect on the Second Syllable

It was observed in our analysis that the initial pitch of the second syllable of the disyllabic word is the pitch of the default tone. In the latter part of the utterance the word the pitch changes according to the tone that the second syllable is linked to.

ga^{+H}hay 'short'

ga^Lhay 'main, head'



Fig. 3.13



Fig. 3.14

gu^{+H}dung] 'hole'

gu^Ldung 'hot'



A close look at the second syllables of the pictures above shows that the second syllable starts at around 110 Hz. If it is a high tone, as in ga^{+H}hay and gu^{+H}dung (figure 3.13 and 3.15), the pitch rises. Whereas in case of a low tone, as in ga^Lhay and gu^Ldung (figure 3.14 and 3.16), the pitch falls downwards from the default mid tone. Whereas medial /h/ is completely voiceless and the pitch of the second vowel takes off from where it left off in the preceding vowel, we notice that voiced stops actually induce a drop in pitch from the first vowel. However, the pitch on the second vowel takes off almost where it ended on the first vowel. More research needs to be done on the consonantal effects on pitch before we can say anything with certainty.

The sonorant onsets in the second syllable do not show any influence on the pitch pattern. The pitch on those syllables systematically rises from the default mid tone. However the onset /s/ in the second syllable with a high tone, shows a peculiar pitch pattern. We observe that /s/ induces high pitch in the second syllable in the context of high tone. As seen in figure 3.17, the onset makes the initial pitch of the second syllable higher, it moves to a higher pitch, as it is associated with the high tone.

hō^{+H}sō 'drive'



Fig 3.17

After the analysis of the tones in disyllables in the previous sections we can postulate that in Bodo disyllables, it is not possible to have tones in both the syllables. In a disyllable there can be only one Tone Bearing Unit and that is always the second syllable. Thus Baro's claim that every syllable of a lexical entry may be specified for tones is not substantiated. Other northeastern languages of the same language family have been reported to show tonal contrasts in both the syllables of a disyllabic lexical entry. But, having re examined Bodo, we feel that, further research in these languages is absolutely necessary before we can make any broad generalization.

The related languages of this family are Bodo, Garo, Dimasa and Tiwa. Among these Burling (2001) shows convincingly that in *Tiwa*, tone is a feature of the whole disyllabic word. According to Burling (2001) Tiwa has a two-tone system and either tone can occur on any syllable. Hence Tiwa words are not only distinguished by the contrastive tones, but also by the position in which they occur. If the tone occurs in the first syllable of the disyllabic lexical entry, tone is spread on to the syllable on the right. If the tone on the first syllable is high then it uniformly induces a high tone across the word. But if the high tone is associated to the second syllable, then the first syllable gets a mid tone or a neutral

pitch. The presence of a similar tonal patterns in most of the northeastern tone languages persuade us to believe that the 'mid' tone found in these languages might actually be a default tone and that in the disyllables we consistently see only a single lexical tone assignment.

If this indeed is the picture, then the areal feature for the language family is that each lexical entry allows only one lexical tone to be assigned. The parametric variations, then, would be:

- a) Bodo, with rightmost lexical tone,
- b) Tiwa, with lexically associated tone,
- c) Whereas Bodo does not allow lexical tone to spread, Tiwa allows a rightward spread of lexical tone.

More research needs to be done before a detailed analysis is possible.

3.3.4. '0' Tone in Disyllables

Our findings till now are contrary to Bhattacharya (1977). Bhattacharya claims that there is a 0 tone in Bodo which copies the tone of the adjoining syllable. He gives certain examples which show that the 0 tone is present in non-derived disyllables. Such as ⁰khu¹ser (sugarcane) and ⁰khu³ser (to prick with nails). However, according to our analysis the first vowel of the disyllabic word always takes the default, mid tone, hence there is no possibility there that any tone is copied from the neighboring syllables. Let us consider ⁰khu³ser first. Ignoring the effect of /r/, we see a low tone with final vowel (figure 3.18). The first vowel carries a mid tone, which exhibits a fall because of the

khu^Lser 'to prick with nails'

khu^{+H}**ser** 'sugarcane'



initial aspirate. Clearly there is no copying of the low tone of the second syllable onto the first. The first syllable has only a default mid tone, contrary to Bhattacharya. Unfortunately, we are not in a position to say much about the so-called '0' pattern in ⁰khu¹ser. The first vowel is too short to carry any tone (figure 3.19), and we assume, once again that there is no question of tone copying. Like the distinction between Tone 2 and Tone 3, Tone 0 could also be redundant.

But if, as Bhattacharya claims, the 0 tone is in the final position of the syllable as it is ¹zen⁰nay[^Mzen^{+H}nay] 'to be defeated' ³zen⁰nay[^Mzen^Lnay] 'to begin'



supposed to be in 1 zen 0 nay and 3 zen 0 nay, what we see is tone transfer, not copy, as shown in the pictures above. In the picture of 1 zen 0 nay we see that the first syllable does not

have a tone, rather it is neutralized with a default mid tone, but the syllable bearing the supposedly 0 tone takes the high tone. We see a rise on the second syllable with the supposedly 0 tone. Similarly in case of 3 zen 0 nay we see a low tone on the second syllable, whereas the tone of the first syllable is neutralized and assigned a default tone.

If we stick on to Bhattacharya's account (ignoring his tonal specifications), we would have the following derivation:

i)	^H zen ⁰ nay	^L zen ⁰ nay	
ii)	zen ^H nay	zen ^L nay	Tone Transfer
iii)	^M zen ^H nay	^M zen ^L nay	Default Mid Tone

However our derivation is simpler. Following an assumption that Bodo words have only one lexical tone, there is no question of tone transfer (we did not find *nay* to be one of the lexical suffixes in the language). What happens is tone association to the rightmost and the insertion of a default tone. Hence we state our derivation as below:

3.3.5 The Tones

In case of monosyllables, we were unable find any distinction between Tone 2 and Tone 3. In our analysis of disyllables we were again unable to gather any spectrographic evidence showing difference between Tone 2 and Tone 3. Contradictory to the claims of Bhattachrya (1977), we see that Tone 2 and Tone 3 are one and the same and it is actually a falling tone. In the pictures below there are not any tangible feature present which can set both of them apart in terms of pitch. From now onwards, we shall not make any distinction between Bhattacharya's Tone 2 and tone 3.



3.4 Conclusion

We arrive at the conclusion that even in disyllabic words we do not see a systematic difference between tone 2 and Tone 3. However, in a disyllabic word also, we assume that a single lexical tone is assigned to the word. Answering the question asked by Burling (2001) about the number of tones assigned to disyllabic words, we could say that only one tone is assigned to the disyllabic word and that too is restricted to the second syllable. The first syllable always takes the default mid tone.

Our conclusion is that every non-derived lexical entry in Bodo is assigned one and only one tonal specification. We then require a rule of tone association to the rightmost syllable and a rule of default mid tone insertion, to apply in that order.

i) [zennay] ^H	[zennay] ^L	
ii) ^H zennay	zennay	Tone Association
iii) ^M ^H zennay	M L zennay	Default Mid Tone Insertion

Though we have not looked at larger sequence of non-derived words, we hope further research will fully confirm our findings.

We must also add that our account now is able to explain the absence of such tonal patterns as given below in non derived words:



Finally, we have also been able to show that Bodo may not have a 0 tone as claimed by Bhattacharya.

Chapter IV

Some Issues in Bodo Morphophonemics

4.1 Introduction

Bhattacharya (1977) shows that, in Bodo, morphology and phonology interacts with each other in an interesting way. Assuming that Bodo has a three way tonal contrast he postulates a regular morphophonemic pattern in Bodo. According to him, the high tone is lowered to the next lower tone (hence, tone 1 > tone 2)¹ in close junctures and low tone is raised to the immediate higher tone (tone 3>tone 2) in close junctures. He claims these alterations to be very regular. Weidert (1987) shows that the high tone in Bodo arises due to a glottal segment in the lexical entry. Hence according to Weidert the high tone in the second syllable of the word /dōikor?/ 'a well' is due to the glottal stop present in the word /dōi?/ 'water'. Therefore when the plural suffix /por/ (transcribed as /phōr/ in our study) is attached to the stem /dōikor?/ we see that the suffix is specified with a high tone as the preceding syllable has a glottal stop at the end. Following Weidert we can represent it as:

4.2 The Hypothesis

Contrary to Bhattacharya and Weidert, we assume a more complex system of morphophonemics in Bodo. We assume that there are at least three different types of tonal patterns in Bodo. (1) In some types of alteration the tone of the stem is neutralized to a mid default tone and the

inherent tonal properties of the clitics are preserved. This alteration can be exemplified as below:

¹ We follow Bhattacharya (1977)'s transcription convention here, though we do not quite agree to his convention of marking the tones in Bodo.

${}^{M}man^{^{+H}}si + {}^{M}g\bar{o}^{^{+H}}bang \rightarrow {}^{M}man^{M}si^{M}g\bar{o}^{^{+H}}bang$

"Man + many \rightarrow many men"

(2) Causative prefixes show another distinct kind of tonal alteration. When the prefix is attached to a root it takes the default tone whereas the tonal specification of the stem is retained. In the following example this type of alteration is shown:

$pho+^{L}thang \rightarrow ^{M}pho^{L}thang$

"pho+sow seeds \rightarrow "to make sow seeds"

(3) Plural suffixes like $-s\bar{o}r$ and $-ph\bar{o}r$ do not have inherent tonal specifications. But when they are attached to a stem, the stem is neutralized. The tonal specification of the stem is transferred to the suffixes. This type of alteration can be exemplified as below:

$^{+H}$ nōng+sōr \rightarrow ^Mnōng $^{+H}$ sōr

"You (hon, singular)+sōr → you (hon, plural)"

(4) Another type of alteration occurs with the plural suffixes, such as $-m\bar{o}n$. In this type the tone of the last segment of the stem spreads to the suffix. The $-m\bar{o}n$ type of alteration can be shown as following:

M nōng^{+H}thang+mōn \rightarrow M nōng^{+H}thang^{+H}mōn

(5) The causative suffix –ho shows another kind of alteration. In this type of alteration both the stem and the suffix retain their inherent tonal specifications.

This affixation can be shown as below:

^{+H}kham+^Lho→^{+H}kham^Lho

"burn+ho \rightarrow "to make burn"

Hence the working hypothesis, we start with in this chapter is as below:

• Bodo derived words will follow the tone assignment pattern of disyllables. In Chapter II we have noticed that in Bodo disyllables the rightmost syllable is specified with a tone, whereas, the initial syllable is assigned a mid default tone (as shown in Figures 4.1 and 4.2).

```
a<sup>+H</sup>doy 'calf of the leg'
```

ga^Lhay 'main, principal, head'



• Since tones are neutralized in non-final syllables we predict the following:

a) under prefixation, the prefix will acquire the default mid tone

- b) under suffixation, there are two possibilities:
 - i) if the suffix is not inherently specified with a tone, the tonal specification of the stem will be transferred to the suffix and the stem will be assigned default, mid tone
 - ii) if the suffix is inherently specified, the tone on the stem will be neutralized to the default mid tone.

We will see in the following section that our working hypothesis is not fully borne out.

4.3 The Findings

4.3.1 Plural Clitics Manōy, Burza and Gōbang

Tonal alteration is observed in the plural clitics like *manōy*, *burza* and *gōbang*. We observed that *gōbang* is specified with a high tone that is retained even if it is affixed to the stem *mansi 'man'*. *Mansi* is specified with a high tone as the pitch picture of its bare form shows in 4.3. When attached to the clitic *gōbang 'many'* the stem *mansi* loses its tonal specification and gets the default tone. Figure 4.4 clearly shows that the tone-bearing unit of the stem *mansi* getting a default tone. The pitch of the default tone is



falling as the stop consonant /g/ makes the pitch lower. The pitch of the clitic $g\bar{o}bang$ is observed to be rising in figure 4.4. This makes us conclude that:

- 1. Clitics have their own tonal specification.
- 2. When attached to words clitics neutralize the tone of the stem, whereas their own tonal specification remain intact. Hence it shows a right aligned tonal specification as in disyllables.

4.3.2 Alteration In Ph-, Bv-, Sv Type of Causative Prefixes

We find that our working hypothesis is correct regarding the tonal pattern of prefix form phV-, bV-, sV type of causative prefixes. The ph- type of causative prefixes includes the prefixes *phi*-,

*phe-, phu-, pha-*and *pho-*². In bV- and sV type of causatives, V is the vowel copied from the first syllable of the stem. *phi-, phe-, phu- and pha-* type of causatives occur if the first syllable of the stem has the vowels /i/, /e/, /u/ and /a/ respectively as nuclease. Examine the tonal specification of the prefix phV-:



^M**phe^Lsem** 'to make something decrease'





^M**phu^Lrung** 'to make something mild'







^Lthao 'be tasteful'

^M**pho^Lthao** 'make something tasteful'

 $^{^{2}}$ Unlike bV- and sV prefixes, the ph- prefix does not attest total copy of the stem vowel. We do not go into the details of the vocalic pattern of the prefix here.



Figure 4.5, 4.7 and 4.9 show that the stems sem, rung and thao are specified for a low tone. When the causative prefix of phV- is added to these stems the tonal specification of the stem does not change. However, the causative prefix takes the default mid tone as shown in figure 4.6, 4.8 and 4.9. It is seen that the default tone in the figures 4.6, 4.8 and 4.10 start at a high pitch. It can be explained with the help of our earlier observation that aspirated consonants always result in the raising of the mid tone.



^M**be**^L**kheo** 'to make somebody open'



Similarly, in figure 4.10 we observe that the stem *geo* is specified with a low tone. When the bV-causative prefix is attached to the stem, the stem does not attest in any tonal alteration and it preserves its inherent tonal specification (figure 4.12).



In figure 4.13 we see that the stem *gubung* is specified for a low tone. When the sV- causative prefix is affixed to the stem, the stem does not undergo any alteration. But the prefix is specified with a default mid tone.

In this type of alteration, we assume that the causative prefixes do not have an inherent tonal specification. But as soon as they are affixed to a stem they acquire a mid and default tone and result in a left-aligned default tonal specification. We conclude that the causative prefix does not have ant lexically specified tone and gets a mid tone by default.

Once again we see that our working hypothesis is vindicated. Notice that according to our hypothesis, prefixes cannot be realized with any tone other that the default, mid tone. Even if they did have a lexically specified tone, the tone on the prefix would always be suppressed and a default, mid tone would replace it. So far our working hypothesis has proved to be correct.

4.3.3 –*s*or and –*ph*or type suffixes

Let us now examine another type of suffixation, e.g. plural suffixes $-s\bar{o}r$ and $-ph\bar{o}r$. Figures 4.27 and 4.29 show this type of affixation.

^{+H}nōng 'you (singular, honorific)'

^Mnōng^{+H}sōr 'you (plural, honorific)'





^Mno^{+H}phōr 'houses'



Figure 4.15 and 4.17 illustrate $n\bar{o}ng$ and no which are specified for a high tone. Figure 4.19 shows *raw* being specified for a low tone. When the plural suffixes $-s\bar{o}r$ and $-ph\bar{o}r$ are affixed to these words, it is observed that the inherent tones of the stems are transferred to the plural

suffixes $-s\bar{o}r$ and $-ph\bar{o}r$. Figures 4.16, 4.18 and 4.20 show that the suffixes are specified with a tone whereas the stems get the default mid tone. In figures 4.16 and 4.18 the suffixes are specified with a high tone as the non-derived stems were specified with a high tone. Whereas the suffix is specified with a low tone in figure 4.20 as the non-derived form of *raw* was specified with a low tone.³

In these cases we see that our working hypothesis b) ii) is vindicated. The suffix has no inherent specification and the specification on the stem gets transferred to the suffix-obeying right alignment. Subsequently, the stem gets a default, mid tone.

The three cases we examined so far are instances of type I derivation in Bodo. Their morphology can be represented as below:

Type I

a.[[stem] suffix]

b. [prefix[stem]]

Tone assignment in this type can be represented as:

³ WE ignore the high tone with segment /r/ in all these cases. (see Chapter II for details)

All these are consistent with our working hypothesis that assumes that derived words are similar to non-derived words in admitting of a single tonal specification on the rightmost syllable.

4.4 The Exceptions

However there two are cases where our hypothesis about the right aligned tone assignment does not hold true. We attempt to describe these cases in the sections 4.4.1 and 4.4.2.

4.4.1 Tthe causative suffix-ho

In this type of suffixation we observe that the suffix is specified for a distinct tonal identity. At the same time the inherent tonal specification of the stem is also preserved. This type of affixation does not result in any tonal alteration of the stem as it is shown in figures below:



^Lkham^Lho 'to make something burn'



^Mpho^{+H}thai 'believe'

^M**pho**^{+H}**thai**^L**ho** 'to make believe'



In figure 4.21 it is observed that the lexical item is specified for a low tone. Though it is affixed with a causative suffix, as shown in figure 4.22, the tonal specification of the stem does not change. In figure 4.23 and 4.25 the words *phothai* and *gi* are specified with high tones. Though the causative suffix -ho is affixed, the tonal specification of the words does not change as the high pitch contours of the words in figures 4.22 and 4.23 indicate.

A comparison of the figures 4.22, 4.24 and 4.26 indicates that the suffix *-ho* shows a distinct tonal specification. Causative suffix *-ho* is seen as being specified for a low tone in all the cases which results in a pitch picture as shown in figure 4.27.



Fig 4.27

Tonologically speaking, the **[stem].[ho]** seems to be made up of the lexical words with distinct tonal specification. It may be the case that the **[stem].[ho]** form are not really single prosodic words but two distinct prosodic words syntactically embedded within the other. This possibility remains to be examined from the morpho-syntactic point of view contrasting the prefixial causative that may function like 'direct causative' and the –ho form that may function like periphrastic causative. This case can be considered as a Type II derivation and its morphology is represented as below:

Type II



If this suggestion turns out to be true, our working hypothesis is still validated.

4.4.2 –mon type alteration

Another alteration is observed in the plural suffix -mon. The pictures below show this type of alteration.



Fig. 4.30



In pictures 4.28 and 4.30 we observe that the lexical items are specified with a high tone. When the $-m\bar{o}n$ suffix is added to them, the high one is spread to the suffix. Figure 4.29 and 4.31 clearly shows that the suffix mon carries the same pitch as of the syllable preceding it. As observed in case of monosyllables and disyllables, the nasal consonant /n/ carries a rising tone.

We assume that the $-m\bar{o}n$ suffix is not specified with an inherent tone. It carries the tonal specification of the preceding syllable. We observe that this kind of alteration is restricted to the stems with a high tone. This case can be considered as a Type III derivation and its morphology is represented as below:

Type III



For the first time we find that there is an instance of tone copying in Bodo. We are not ready to advance any hypothesis regarding tone copying in Bodo till a large number of suffixes of different types namely, derivational and inflectional have been examined in detail.

4.5 Conclusion

We conclude that in Bodo derived words we observe a tonal pattern that is similar to the tone assignment pattern in disyllables derived words too, like the disyllabic lexical entries the tone assignment tends to be right aligned.

A wild hunch that we have is the following. Following the assumption in Lexical Phonology (Kiparsky, 1985), if we assume that these are two distinct levels of word formation in Bodo, namely level I and the word level, we may find that Level I usually behaves like non-derived words (as in English) where every 'word' has a single tonal specification but word level affixation may relax the strict phonotactics of the earlier level (again as in English-contrast non-geminate 'innumerable' with double 'unnecessary') allowing tone copy.

We saw that prefixal causative and ' $g\bar{o}bang$ ' (see section 4.3.1) pluralization mimic non-derived words in permitting only a single tone specification on the rightmost syllable. Further, $-ph\bar{o}r$, -

 $s\bar{o}r$ pluralization results in transferring tonal specification on the stem to the suffixes, neutralizing the tone on the stem, once again resulting in output form obeying the phonotactics of non-derived words. We feel that these processes may be assigned to Level I of Bodo lexical phonology.

The lone case of $-m\bar{o}n$ pluralization illustrates the process of tone copying in Bodo. If it can be shown that $-m\bar{o}n$ pluralization is distinct from other types of pluralization, not only in admitting multiple tones within a word but also other morphological features like attachment to compound noun etc. A slender piece of evidence with us is data like:

[[lai] [siri]_{feminine}] mon.

If this suggestion is correct, then we could assign $-m\bar{o}n$ affixation to the word and allow tone copying at this level. However, we need to examine a lot more cases of suffixation before we can come to a conclusion.

Finally looking at the multiple tone specification in [*ho*] causativization, it would fit in nicely with our model of Bodo morpho-phonemics if it could be shown that the [ho] system is really an instance of syntactic, periphrastic causativization.

The morpho-syntactic representations of the three types we have examined are the following:

Type I



57

Word level

Type II T [[stem] suffix] prwd prwd

Type III T T [stem] . [suffix] prwd prwd

Chapter V

An OT Analysis of Bodo Tones

5.1 Introduction to Optimality Theory

OPTIMALITY THEORY (Prince and Smolensky 1993, McCarthy and Prince 1993a,b) is a model of Generative Grammar, which is non-derivative. In OT the output is the simultaneous application of EVAL to the input. The relation between inputs and outputs is mediated by two formal mechanisms GEN and EVAL. GEN assigns possible structure to the input and EVAL applies the constraint hierarchy to select the best candidate among those created by GEN. The only function of the grammar of a language is to rank and order the constraints, which belong to the UG. To treat the variations across languages in the world, OT assumes that the universal constraints are re-ranked by the grammar of different languages. Earlier theories assumed that variations across languages were the result of parametric selection. OT, unlike parametric theories, asserts that all constraints are present in all languages, the only difference being in the ranking of the constraints.

OT has a set of constraints called the **Faithfulness Constraints**, which tries to preserve the input forms. To preserve different aspects of the input form, OT has different types of **Faithfulness Constraints**. However the desire to preserve the input form varies from language to language. That variation can be taken care of by the ranking of the **Faithfulness Constraints**. On the other hand **Markedness Constraints** use crosslinguistic evidence to avoid specific features or structures. These constraints account for the segmental inventory, syllable structure and phonological alternation, in short, any aspect of linguistic phenomena, be it phonological, morphological or syntactic. The **Faithfulness Constraints** see to it that the specification of the input is preserved in the output, whereas **Markedness Constraints** try to select candidate representation that decrease the markedness of the representation. With the help of constraint ranking, among many outputs the best one is selected as the most plausible output. OT assumes that selected output is optimal. OT evaluates an infinite set of candidate output forms generated by GEN on an input. The winning candidate is the optimal one as it incurs least serious violations among a set of constraints.

Constraint ranking and the selection of the most suitable output are displayed in Table 1. A key to the table is given below.

- * = violation
- ! = fatal violation
- *^{ce}* = the most suitable candidate
- Shaded cells no longer matter because a higher ranked constraint has made the decision.

C ₁ is a higher-ranked constraint than C ₂				
	C ₁	C_2		
a. 🖙 candidate a		*		
b. <i>candidate b</i>	*!			
Table 1				

 $C_1 >> C_2$ C_1 is a higher-ranked constraint than C_2

As shown in Table 1, a tableau lists the outputs vertically in any order. The constraints are listed horizontally. As shown in Table 1, candidate a and candidate b are two possibilities (among the infinite set) generated by GEN. A solid line separates the constraints C1 and C2 indicating strict domination. *Candidate a* satisfies constraint C₁ but violates constraint C₂ (indicated by a '*'). However *Candidate b* violates constraint C₁ and may or may not satisfy constraint C₂. The violation or satisfaction of constraint C₂
does not matter anymore, as C_2 is a lower ranked constraint; moreover the higher ranked constraint C_1 has already made the choice clear. The constraint hierarchy assumes that the violation of C_1 is much more serious (indicated as '*!') than the violation of C_2 . The violation of C_2 is irrelevant if C_1 is violated. Hence *Candidate B* cannot emerge as a suitable output (as it violates constraint C_1). Therefore *Candidate A* emerges as the optimal output as indicated by a '@'' (even though it may violate C_2).

5.2 OT Treatment of Tones

Yip (2002) is an attempt at an OT analysis of tonal phonology. She proposes a few constraints pertaining to tone. She lists a few modifications of the well-formedness conditions about tone proposed by Goldsmith (1976):

- Tones are usually associated with syllables, but not always
- Syllables are usually associated with tones, but not always
- Association is preferably one-to-one, but not always
- Tone (especially H tone) is attracted to prominent positions (beginnings of things, edges, accented or stressed syllables) but not always.

All these can be true in some but not all languages. Each of these can be stated as markedness constraints. They are expressed by Yip (2002) as follows:

• ***F**LOAT: A tone must be associated with a Tone Bearing Unit (TBU).

This constraint makes sure that an output like the following is not generated where tone T^3 is not associated with a TBU:

$$\begin{pmatrix} \sigma & \sigma \\ | & | \\ T^1 & T^2 & T^3 \end{pmatrix}$$

• **SPECIFY T**: A TBU must be associated with a tone.

This constraint rejects an output like the following, as the second syllable is not specified with a tone:

$$\begin{pmatrix} \sigma & \sigma & \sigma \\ | & | & | \\ T^1 & T^2 \end{pmatrix}$$

• NoContour: A TBU may be associated with at most one tone.

This constraint rejects an output where a TBU is associated with more than one tone:

$$* \left(\begin{array}{c} \sigma \\ \\ \\ \\ T_{T} \\ 1 \end{array} \right)$$

• NoLong T: A tone may be associated with at most one TBU.

Hence an output like the one shown below is to be strictly avoided:

$$\left(\begin{array}{c} \sigma & \sigma & \sigma \\ I & I & I \\ T & I \end{array}\right)$$

• ALIGN-TONE: Align the specified edge (L/R) of a tone span with the head or edge (L/R) of a prosodic or morphological unit.

For example, when this constraint prefers the rightmost syllable of a language to be specified with a tone it would prefer only the following structure:

$$\begin{pmatrix} \sigma & \sigma & \sigma \\ & & | \\ & & T \end{pmatrix}$$

In non-derived lexical items Bodo prefers this structure as the tone is linked to the rightmost syllable:

$$\begin{pmatrix} go & ba \\ & | \\ & T \end{pmatrix}$$

Apart from these markedness constraints there are some general faithfulness constraints that preserve underlying contrasts of tone quality and placement:

Tonal faithfulness constraints

• **DEP-T**: No insertion of tones.

This constraint restricts the insertion of a new tone in the output form. If a new tone is inserted then the output is considered to be violating this constraint.

• MAX-T: No deletion of tones.

This constraint prevents the deletion of a tone present in input in its output form.

• *Associate: No new association lines.

This constraint restricts a tone from specifying a new TBU in the output. It makes tones to stay in their original position.

• *DISASSOCIATION: No removal of association lines.

This constraint makes it sure that a tone association stays in its original position. It prevents a tone from disassociating with the TBU it specifies.

• NoFusion: Separate underlying tones must stay separate.

Hence, two or more tones cannot come together and specify a single TBU.

• **IDENT-T**: Correspondent tones are the same.

The correspondence of tones in the output is as it is in the input.

• LINEARITY: Preserve underlying linear order.

The order in which tones occur in the output is the same as it is in the input.

Yip attempts at capturing Goldsmith's observations about the preference for contours and plateaux at the right edge of the word. Goldsmith calls it left-to-right association. In OT Yip captures Goldsmith's observation with the help of alignment constraints like the ones shown below:

• ALIGN-L: Each T should align with the left edge of the domain (gradiently assessed)

This constraint prevents a tone from occurring anywhere except the left edge of the word so that an output like

$$\begin{pmatrix} \sigma & \sigma & \sigma \\ | & \\ T & \end{pmatrix}$$
 is obtained, but not like
$$\begin{cases} \sigma & \sigma & \sigma \\ | & \\ T & \\ \end{array}$$

• ALIGN-R CONTOUR: Contour tones should align with the right edge of the domain. Therefore in Mende

 $\begin{pmatrix} ny\hat{a} h\hat{a} \\ & & \\ T & T \\ T & T \\ \end{pmatrix} is possible, but \begin{pmatrix} * & ny\hat{a} h\hat{a} \\ & & \\ & & \\ T & T \\ \end{pmatrix} is not possible.$

According to Yip, tone is also subject to more general phonological conditions such as the Obligatory Contour Principle (OCP), locality, and markedness constraints:

• **OCP**: Adjacent identical elements are prohibited.

Leben (1973) proposed the Obligatory Contour Principle (OCP), which says that words with sequences of high toned syllables must be represented as in (a), not as in (b):

a.
$$\sigma \sigma$$
 σ NOT * b. $\sigma \sigma$ σ
H H H H

Later it was found that this principle covers not only high tone but also other tones. Its plausibility as a constraint on outputs made its way to OT.

• NoGAP: Multiply linked tones cannot skip TBUs.

A set of TBUs, which are linked by only one tone cannot leave a TBU in the middle unspecified with a tone. Hence,

$$\begin{pmatrix} \sigma & \sigma & \sigma \\ H \end{pmatrix}$$
 is possible, but not
$$\begin{pmatrix} \sigma & \sigma & \sigma \\ H \end{pmatrix}$$

• LOCAL: Spread only to the adjacent items.

When an association changes the new association line is formed associating the adjacent item. Hence, for the input

$$\begin{pmatrix} \sigma \sigma \sigma \sigma \sigma \\ H \end{pmatrix}$$
 the output can be
$$\begin{pmatrix} \sigma \sigma \sigma \sigma \sigma \\ H \end{pmatrix}$$
 and not
$$\begin{pmatrix} \sigma \sigma \sigma \sigma \sigma \\ H \end{pmatrix}$$

• General markedness: *H>>*L

This constraint shows the universal preference for a low tone in comparison to a high tone. High tone is more marked than low tone.

It is worth mentioning that in OT all the constraints are universal and present in the grammars of all languages. If a constraint is very low ranked it is assumed that its effects are not visible and hence will not be discussed. Nevertheless it is to be assumed that the constraints exist in all languages.

5.3 OT Treatment of Bodo Tones

5.3.1 Introduction

In this section we use OT to account for the tonal phenomenon in Bodo. Assuming the facts from Chapter II, III and IV, we identify some constraints from the ones suggested by Yip (2002), necessary for OT treatment of tones in Bodo:

Tonal faithfulness constraints

- **D**EP-**T**: No insertion of tones.
- MAX-T: No deletion of tones.
- *Associate: No new association lines.
- ***D**ISASSOCIATION: No removal of association lines.
- NoFusion: Separate underlying tones must stay separate.
- **IDENT-T**: Correspondent tones are the same.
- **LINEARITY**: Preserve underlying linear order.

Tones also take into account more general phonological conditions like the Obligatory Contour Principle (OCP), locality, and markedness constraints. These conditions are involved in the tonological analysis of Bodo. The conditions that we take into account for Bodo are as follows:

Tonal markedness constraints

- **OCP**: Adjacent identical elements are prohibited.
- NoGAP: Multiply linked tones cannot skip TBUs.
- LOCAL: Spread only to the adjacent items.
- ***FLOAT:** A tone must be associated with a TBU.
- **SPECIFY T**: A TBU must be associated with a tone.
- NoLong T: A tone may be associated with at most one TBU.
- ALIGN-TONE R: Each T should align with the right edge of the domain.

5.3.2 A non-derived lexical item must be specified with a tone

In Chapter II and III we saw that all Bodo non-derived lexical items underlyingly carry a single lexical tone. They are specified with either an H (high) or an L (low) tone. Broadly, it can be assumed that in case of Bodo non-derived lexical items the TBU must retain the lexical tone in the output. It is not possible to have a non-derived lexical entry without being specified for a lexical tone. Hence we propose the constraint **SPECIFY (LT, PRWD)** which says that a prosodic word must be specified for lexical tone.

We assume that Bodo non-derived lexical items are underlyingly specified with a tone. Therefore to rule out the possibility of a high or low tone being inserted, we propose the constraint **D**EP-LT (**H**, **L**). This constraint restricts new High or Low lexical tone insertion into a non-derived lexical item. Hence the following situations are not possible:



In Chapter III we observed that the underlying high or low tone, with which Bodo disyllables are specified, is right aligned. Further in Chapter IV we observed that the right edge of a derived polysyllabic lexical entry is specified by a lexical tone. Hence we propose the constraint **ALIGN-R** (**PRWD**, LT) that makes sure that the right edge of the domain in a prosodic word is aligned with the underlying tone. It prohibits the following situations:

a. *
$$\begin{pmatrix} \sigma & \sigma \\ | & | \\ T_i & T_j \end{pmatrix}$$

b. * $\begin{pmatrix} \sigma & \sigma \\ | \\ T \end{pmatrix}$
c. * $\begin{pmatrix} \sigma & \sigma & \sigma \\ | \\ T_i & T_j \end{pmatrix}$

The OT analysis therefore rules out any possibility of a lexical item being specified with a lexical tone, anywhere except the right edge of the domain. Hence an input, which, for example, has its left edge specified with an underlying tone, loses its left edged tonal specification in the output in Bodo. This results in the violation of the tonal faithfulness constraint ***DISASSOCIATE**, which restricts the removal of association lines. As Bodo allows this violation, ***DISASSOCIATE** is considered to be a low ranked constraint compared to the other constraints. Following is the violation of this constraint:

$$\begin{pmatrix} \sigma & \sigma \\ | & | \\ L^T L^T \end{pmatrix} \longrightarrow \begin{pmatrix} \sigma & \sigma \\ L^T L^T \end{pmatrix}$$

- a) SPECIFY (LT, PRWD): A prosodic word must be associated with a tone
- b) ALIGN-R (PRWD, LT): Each right edge of the domain should align with the T
- c) **DEP-LT** (H, L): No insertion of High and Low lexical tones

>>

d) *DISASSOCIATE: No removal of association lines

The ranking logic among individual constraints is of the following nature:

SPECIFY (**LT**, **P**_{**RWD**}) >> ***D**_{**I**SASSOCIATE ; the input has to lose certain association lines in the output, if all the TBUs are specified with a tone each. In that case the output would}

violate ***D**ISASSOCIATE, yet the output would be plausible. Hence the constraint **SPECIFY** (**LT**, **P**RWD) dominates the constraint ***D**ISASSOCIATE.

ALIGN-R (LT, PRWD)>> ***DISASSOCIATE**; if any other TBU than the right aligned one is specified with a tone in the input, the association lined of those TBUs have to be removed. Hence **ALIGN-R** (PRWD, LT) dominates the constraint ***DISASSOCIATE**.

We assume the following ranking for the constraints a), b), c) and d):

Specify (LT, Prwd), Align-R (prwd, lt), Dep-LT (H, L) >> *Disassociate

INPUT σσ LTLT	SPECIFY(LT, PRWD)	ALIGN-R (PRWD, LT)	DEP-T (H, L)	*DISASSOCIATE
а. б б LT		*!		*
b.σ σ LT LT		*i		
C. σσ	*!	*!		**
d. σ σ @ T				*

Specify (LT, Prwd), Align-R (prwd, lt), Dep-LT (H, L) >> *Disassociate

Table 2

The analysis above shows that each syllable of the input is specified with a tone. The input has at least four possible candidates among others. *Candidate a*) is specified with a tone only on the left edge. Hence, it violates the constraint **ALIGN-R** (**PRWD**, **LT**). As this constraint is a higher-ranked constraint, its violation is considered to be a serious one. Hence *Candidate a*) fails to be the optimal candidate in the analysis. Nevertheless it fulfills the constraints **SPECIFY** (**T**, **PRWD**) and **DEP-LT** (**H**, **L**). It violates the constraint ***DISASSOCIATE**, but as a higher-ranked constraint (**ALIGN-R** (**PRWD**, **LT**)) has already decided the choice, its violation is no more significant for the analysis.

Candidate b) violates **ALIGN-R** (**PRWD**, **LT**) as one of the tones is not right aligned. However it fulfills the other two constraints- **DEP-LT** (**H**, **L**) and ***DISASSOCIATE**.

Candidate c) is not specified with a lexical tone. Hence it violates the constraints Specify

(**T**, **PRWD**) and **ALIGN-R** (**PRWD**, **LT**). There is a double violation of the constraint ***DISASSOCIATE**, as the tones of the syllables are removed twice in the output. It fulfills the constraint **DEP**-**LT** (**H**, **L**).

Candidate d) satisfies the high-ranking constraints. However it does not satisfy the constraint ***DISASSOCIATE**. As the high ranked constraint has already made the choice, the violation of the dominated constraint ***DISASSOCIATE** is not significant for the analysis. Hence *Candidate d*) emerges as the winner in this analysis as indicated by a " \mathfrak{P} ". An example from Bodo makes this analysis clear:

INPUT ga hay T T	SPECIFY (T, PRWD)	ALIGN-R (PRWD, LT)	DEP-T (H, L)	*DISASSOCIATE
a. ga hay T		*!		*
b. ga hay T T		*!		
c. ga hay	*!	*!		**
d. ga hay ☞ │ T				*
		Table 3		

Specify (T, Prwd), Align-R (prwd, Lt), Dep-LT (H, L) >> *Disassociate

The explanation given for table 2 applies to table 3 as well.

5.3.3 A TBU without a lexical tone is specified with a default mid tone

In the previous chapters we found some spectrographic evidence to say that Bodo has an underlying default mid tone. We noticed in non-derived monosyllables that the pitch track for a tone always begins at a particular frequency. We identify this frequency as the frequency for mid tone. In the picture below we see that the pitch track begins at 120-125



Hz. Hence this frequency can be identified as the frequency for a default mid tone.

Again in disyllables we noticed that only the rightmost TBU, i.e. the syllable is specified with a lexical tone (L or H). The other TBUs are specified with the default mid tone. In figure 5.2 we see that the disyllabic non-derived lexical item /erso/ is specified with a



Fig 5.2

lexical tone (High) on the rightmost syllable. The first or the leftmost syllable is specified with a default mid tone.

It is observed that no TBU in a lexical item in Bodo can be left without being specified with a tone. The TBUs either have to be specified with a lexical or a default tone. These facts prompt us to suggest a few new constraints (suitably modified) that can account for the occurrence of default, mid tones in Bodo, apart from the constraints proposed in the previous section.

The fact that every TBU in Bodo has to be associated with a tone, prompts us to introduce the tonal markedness constraint **SPECIFY T.** This constraint states that a TBU must be associated with a tone. In Bodo this constraint is ranked as an undominated constraint. We propose a new constraint **ALIGN-L** (**DT**, **PRWD**), which makes sure that the leftmost TBU in a prosodic word is specified with a default tone (DT). Considering the consistency noticed in its occurrence in Bodo, this constraint is considered to be a high-ranked one. It however does not rule out the following possibility in Bodo:

a. *
$$\begin{pmatrix} \sigma & \sigma \\ \downarrow \cdot \cdot \cdot \cdot \\ DT \end{pmatrix}$$

This possibility is ruled out by the high-ranking constraint ALIGN-R (PRWD, LT). Here **a.** is not possible as the default mid tone spreads to the rightmost syllable too. Ideally every lexical item must be specified with a lexical tone on the right and the default mid tone must be specified only on the left of the prosodic word. In **a.** the lexical item is not specified with any lexical tone. Hence it violates the high-ranked constraint SPECIFY (LT, PRWD). SPECIFY (LT, PRWD) makes sure that a prosodic word is specified with a lexical tone. As it does not happen in **a.**, it leads to the violation of the constraint SPECIFY (LT, PRWD). But consider **b.** below:

b. *
$$\begin{pmatrix} \sigma \\ | \\ LT \end{pmatrix}$$

b. is not possible, as we have seen in figure 5.1 that even in monosyllables, the initial part of the TBU is specified with a default mid tone. This prompts us to present a possible case of tone assignment in monosyllables as below:

The markedness constraint **NoContour** is dominated in Bodo by **ALIGN-R** (**PRWD**, LT). This constraint generally restricts a TBU from being specified with more than one tone. But **c.** is possible in Bodo monosyllables, as **ALIGN-R** (**PRWD**, LT) is higher-ranked than **NoContour**.

$$\mathbf{d.} * \left(\begin{array}{c} \boldsymbol{\sigma} & \boldsymbol{\sigma} \\ \boldsymbol{\sigma} \\ \boldsymbol{\sigma} \\ \boldsymbol{\sigma} \\ \boldsymbol{\sigma} \\ \boldsymbol{\sigma} \end{array} \right)$$

In **d.** the leftmost syllable of the prosodic word is not specified with a tone. Hence it violates the high-ranked constraint **SPECIFY T.** Moreover the default mid tone is specified on the rightmost syllable of the prosodic word. Hence d. also violates the constraint **ALIGN-L (DT, PRWD)**.

We also propose a new constraint **D**EP- DT, which prohibits the insertion of a default tone in the output. Ideally this constraint should restrict the following derivation:

$$\begin{pmatrix} \sigma & \sigma \\ | \\ LT \end{pmatrix} \longrightarrow \begin{pmatrix} \sigma & \sigma \\ | & | \\ DT & LT \end{pmatrix}$$

Nevertheless in Bodo, default tone is inserted wherever necessary. Hence, this constraint is considered to be a dominated or a low ranked constraint.

The constraint **No Long LT** restricts a lexical tone from being associated with more than one TBU. It is considered to be a high-ranked constraint in the present analysis. Hence the following output is not possible:



Another constraint **No Long DT** is proposed, which restricts a default tone from being associated with more than one TBU. However this is regarded as a dominated constraint as in case of trisyllables and derived lexical items a default tone is observed to have spread to the nearby TBUs. Hence the following output can be optimal in Bodo:

Again a possibility like the following is ruled out in Bodo:

$$\begin{pmatrix} \sigma & \sigma & \sigma \\ 0 & & & \\ DT & L \end{bmatrix}$$

The high-ranking constraint **No Long LT** prevents the spreading of the lexical tone. Moreover another high-ranking constraint **ALIGN-R** (**PRWD**, **LT**) prevents anything but the right of the prosodic word from being associated with the lexical tone.

The new constraints proposed in this section can be listed as below:

- a) SPECIFY T: A TBU must be associated with a tone.
- b) ALIGN-L (DT, PRWD): Each default tone should align with the left edge of the domain,
- c) **DEP-DT**: No insertion of default tones,

- d) No Long LT: A lexical tone must be associated with at most one TBU,
- e) No Long DT: A default tone must be associated with at most one TBU.
- f) NoContour: A TBU may be associated with at most one tone.

Hence the constraint ranking for Bodo monosyllables is as following:

SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD), DEP-LT (H, L)

		D	bibbboeinii		OOK, DH = 1	01		
INPUT	SPECIFY T	SPECIFY (LT, PRWD)	ALIGN-R (PRWD, LT)	ALIGN-L (DT, PRWD)	DEP-LT (H, L)	*DISASSOC IATE,	NOCONTO UR	DEP- DT
a. G LT				*!				
b. σ DT		*!	*!			*		
c. σ	*!	*!	*!	*!		*		
[☞] d. σ ∧ DT LT							*	*

	>>		
*DISASSOCIATE,	NoContour,	DEP- D	Т

Table 4

In the table above, *candidate* **a.** cannot be the optimal candidate as it violates the high-ranking constraint **ALIGN-R** (**PRWD**, LT). *Candidate* **b.** violates the high-ranking constraint **SPECIFY** (**LT**, **PRWD**) as the candidate does not have any TBU specified with a lexical tone. *Candidate* **c.** is not specified with any tone. Hence it violates the constraints **SPECIFY T**, **SPECIFY** (**LT**, **PRWD**), **ALIGN-R** (**PRWD**, LT) and **ALIGN-L** (**DT**, **PRWD**). *Candidate* **d.** satisfies all the high-ranking candidates. However as **d.** is specified with two tones, it violates the constraint **NOCONTOUR**. As **NOCONTOUR** is a dominated constraint, its violation does not affect in the emergence of **d.** as the optimal candidate.

Tone assignment in the Bodo monosyllable /khon^H/ can be shown with the help of the following optimality analysis:

INPUT khon H	SPECIFY T	SPECIFY (LT, PRWD)	ALIGN-R (PRWD ,LT)	ALIGN-L (DT, PRWD)	DEP-LT (H, L)	*DISASSOC IATE,	NOCONTO UR	DEP- DT
a. khon								
Н			 	*!			 	
b. khon					1 1 1			
M		*!	*!			*		
c. khon								
1	*!	*!	*!	*!		*		
۳d. khon								
∧ м н							*	*

SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD), DEP-LT (H, L) >>*DISASSOCIATE, NOCONTOUR, DEP-DT

Table 5

The analysis presented in table 4 is applicable to table 5.

The constraints suggested in this section, for optimality analysis of Bodo, can be ranked as below to find out the optimal candidate in Bodo polysyllables:

			, = =- ;						
INPUT σσσ LT	SPECIFY T,	SPECIFY (LT, PRWD)	ALIGN- R (LT, PRWD)	ALIGN-L (DT, PRWD)	DEP-LT (H, L)	NO LONG LT	*DISASS OCIATE	DEP- DT	NO LONG DT
a.σ σ σ LT	**!			*!					
b.σσσσ DT LT	*!			*!				*	
C.σσσ DT LT	*!							*	
d.o o o \ DT LT						*!			
e. σσσ ☞ ℓ ∣ DT LT								*	*

SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), DEP-LT (H, L), ALIGN-L (DT, PRWD), NO LONG LT>>*DISASSOCIATE, DEP- DT, NO LONG DT

Table 6

The analysis in table 6 shows that in the input only the rightmost syllable of the word is specified with a tone, i.e. a lexical tone. The other two syllables are not specified with any tone. This results in the violation of the constraint **SPECIFY T** which says that each TBU must be associated with a tone. As the first two syllables of the word are not specified with tones, it violates **SPECIFY T** twice. As this constraint is a high-ranking constraint, its violation makes *candidate a* implausible.

In *Candidate b*) the second syllable of the lexical item is specified with a default tone. However the leftmost syllable is left unspecified for a tone. This results in the fatal violation of the constraint **ALIGN-L** (**DT**, **PRWD**), which says that DT or default tones should be left aligned. As a default tone is inserted in the output, *candidate b*) violates the constraint **DEP- DT** which restricts the insertion of default tones. However, the constraint **DEP- DT** is a low ranking one. Hence its violation does not matter. But the violation of the high-ranked constraint, **ALIGN-L** (**DT**, **PRWD**) has already made *candidate b*) non-optimal.

In *Candidate c*) the syllable on the left edge is specified with a default tone. Therefore it satisfies the constraint **ALIGN-L** (DT, PRWD). However the second syllable of the lexical item 4 is not specified with any tone, which results in the violation of the constraint **SPECIFY T**. As **SPECIFY T** is a high-ranked constraint, its violation makes *candidate c*) non-optimal. This candidate also violates **DEP-DT**. As the high ranked constraint **ALIGN-L** (DT, PRWD) has already made the choice the violation of the low ranked constraint **DEP-DT** is irrelevant. In case of *candidate d*), one lexical tone is associated with two TBUs. The constraint **No Long LT** says that a single lexical tone cannot be associated with more than one TBU. Hence *candidate d*) violates the constraint **No Long LT**. As **No Long LT** is a higher

ranked constraint, its violation results in the rejection of *candidate d*) as a non-optimal candidate.

In *Candidate e*) the default tone is associated with the leftmost syllable of the domain and it spreads to the adjoining syllable, which also is not specified with a lexical tone. This candidate fulfills all the higher ranked constraints but violates the dominated constraints **DEP-DT** and **No LONG DT**. As in the output, default tone is inserted; the candidate violates the constraint **DEP-DT**. As the tone is spread from the leftmost syllable to the adjoining syllable, it violates the constraint **No Long DT**. This constraint restricts default tones being associated with more than one TBU. However as these two constraints are dominated, their violation can still make the candidate plausible. Therefore *candidate e*) emerges as the optimal candidate in the analysis.

Here one might posit the question why not use the tonal faithfulness constraint **OCP** which prevents adjacent identical items, to rule out possibilities like:

$$\begin{pmatrix} \mathbf{\sigma} & \mathbf{\sigma} \\ | & | \\ T_i & T_i \end{pmatrix}$$

Our answer to this problem would be that with the help of other existing high-ranked constraints, such as the ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD) and DEP-LT (H, L), such possibilities are anyway ruled out. A candidate like the following is ruled out with the existing constraint



ALIGN-R (PRWD, LT). This constraint says that only the right edge of the prosodic word should be specified with a lexical tone. Moreover the constraint **DEP-LT** (**H**, **L**) prevents the insertion of lexical tone. As a lexical tone is inserted in this derivation, **DEP-LT** (**H**, **L**) is violated which makes this derivation non-optimal.

As the high ranked constraint **ALIGN-L** (**DT**, **PRWD**) restricts default tones (DT) only to the left edge of the prosodic word, a possibility like the following is ruled out:



As shown here, the existing constraints very well make the choice of the optimal candidate. Hence, **OCP** does not perform any additional work for us. However, since **OCP** is deemed to be a universal constraint we assume that it is redundantly present in the grammar of Bodo as well. But we choose to ignore it in the present analysis.

An example from Bodo is given in table 7 to show the viability of the constraint ranking for Bodo non-derived polysyllabic entries:

		*	^c Disassoc	лате, Def	?- dt, No I	LONG DT			
INPUT ga hay I I LT	SPECIFY T,	SPECIFY (LT, PRWD)	ALIGN- R (PRWD, LT)	ALIGN-L (DT, PRWD)	DEP-LT (H, L)	NO LONG LT	*DISASS OCIATE	DEP- DT	NO LONG DT
a.ga hay	*!			*!					
b. ga hay LT LT			*!		*!				
c. ga hay ^{CP} DT LT								*	
d. ga hay DT DT		*!	*!	*!			*	**	
				Tab	e 7				

SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), DEP-LT (H, L), ALIGN-L (DT, PRWD), NO LONG LT

>>

In the analysis above the input /gahay/ is specified with a lexical tone on the right edge of the prosodic word. Candidate a) in the analysis violates the constraint SPECIFY T as the leftmost TBU of the prosodic word is not specified with a tone. As SPECIFY T is a higherranked constraint, its violation makes *candidate a*) an implausible output.

Candidate b) is specified with a lexical tone each on both the syllables of the prosodic word. It also violates the high-ranked constraint ALIGN-R (PRWD, LT). The constraint ALIGN-R (PRWD, LT) restricts the lexical tones to the rightmost TBU of a prosodic word. As the leftmost TBU is also specified with a lexical tone, the candidate violates the constraint ALIGN-R (PRWD, LT). The insertion of a lexical tone on the left edge of the prosodic word violates the constraint DEP-LT (H, L). This DEP constraint prevents the insertion of a lexical tone. SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT) and DEP-LT (H, L) are all undominated, high-ranked constraints. Hence their violation makes *candidate b*) an implausible output.

Candidate c fulfills all the high-ranked constraints. Nevertheless it violates a low ranked constraint **DEP- DT**. As **DEP- DT** is a dominated constraint, its violation can still make *candidate c*) the optimal output in the absence of a more suitable candidate.

Candidate d) is specified with a default tone on both of its TBUs. As there is no lexical tone associated with any of the TBUs, it violates the constraint **SPECIFY (LT, PRWD)**. Apart from this, the second instance of DT, which is linked to the right syllable, violates **ALIGN-L (DT, PRWD)**. As the violated constraints **SPECIFY (LT, PRWD)** and **ALIGN-L (DT, PRWD)** are high-ranked, *candidate d* does not emerge as an optimal output.

5.3.4 Tone assignment in Bodo derived Polysyllables

We now consider derived words and their tone pattern in Bodo. Though Bodo derived polysyllables follow the system of tone assignment as described in section 5.3.2, we observed some cases that do not follow the general rule of tone assignment. The following sections give an optimality account of the default tone assignment patterns in derived polysyllables along with the tone assignment patterns in $-m\bar{o}n$ and -ho type of suffixes.

5.3.4.1 The general pattern

In Chapter IV we discussed the general tone assignment pattern of Type I derived polysyllables. We noticed that Type I derivations in Bodo follow the tone assignment pattern of the non-derived polysyllabic lexical entries allowing us to assign the word formation process to level I of Bodo lexical phonology model (Kiparsky, 1985). We saw three possibilities in this type, as following:



In all these cases the derived form follows the tone assignment pattern of the non-derived polysyllabic entries, where the lexical tone is right aligned. If there is a lexical tone that is not right aligned, it is either transferred to the right of the prosodic word or deleted if the right of the lexical item is already specified with a tone. In case of (a) type of derivations, we see that the tone of the suffix is preserved whereas the tone of the stem is lost. This prompts us to arrive at the conclusion that **Max Sf Lt>> Max Stm Lt** is underlyingly present in Bodo derivations. An example each for the a, b and c type of derivations can show how in the OT analysis of Type I derivations, the constraint hierarchy suggested for non-derived lexical entries will select the desired representation as optimal.

5.3.4.1.1 Type I (a) derivation

M man $^{+H}$ si $+^{M}$ go $^{+H}$ bang $\rightarrow ^{M}$ man M si M go $^{+H}$ bang

Here, the derived word is considered to be a prosodic word and tone assignment is carried out accordingly. The constraint hierarchy for this type of derivation is as following:

MAX SF LT, SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), DEP-LT (H, L), ALIGN-L (DT, PRWD), NO LONG LT

>> *Disassociate, Dep- dt, No Long DT, Max Stm Lt

INPUT [[mansi]gōbang] ^{prwd} M H M H	MAX SFLT	SPECIF YT,	SPECIF Y(LT, PRWD)	ALIGN- R(LT, PRWD)	ALIGN- L(DT, PRWD)	DEP-LT (H,L)	NO LONG LT	*DISAS SOCIAT E	DEP- DT	NO LONG DT	MAX SIMLT
a. [[mansi]gōbang] ^{prwd} M H M H				*!	*!						
b. [[mansi]gōbang] ^{prwd} M M M H					**!			*	*		
C. [[mansi]gōbang] ↓ ^{prwd} ☞ M H								**		**	*
d. [[mansi]gōbang] k	*!							**		**	

MAX SF LT, SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), DEP-LT (H, L), ALIGN-L (DT, PRWD), NO LONG LT>>*DISASSOCIATE, DEP-DT, NO LONG DT, MAX Stm LT

Table 8

In the analysis above, the input is considered as a single prosodic word. It violates the high ranking constraint **ALIGN-L** (**DT**, **PRWD**), as the default M(id) tone shows up in the third syllable of the prosodic word. This candidate also violates the high-ranking constraint **ALIGN-R** (**PRWD**, LT). This constraint makes sure that the lexical tone is associated with the right of the prosodic word. Hence, this candidate cannot be an optimal output.

Candidate b violates **ALIGN-L** (**DT**, **PRWD**) as apart from the leftmost syllable, there are two more TBUs, which are individually specified with a default tone. It violates the high-ranking constraint **ALIGN-L** (**DT**, **PRWD**) twice. As in this analysis a default tone is inserted, the low-ranking constraint **DEP-DT** is violated.

Candidate c does not violate any of the high-ranking constraints. Nevertheless, it violates the low-ranking constraints **No Long DT.** It violates this constraint as the default tone is

spread to the other two syllables. However, its violation is not serious in nature and the candidate emerges as the optimal output.

Candidate d is non optimal, as it violates the high-ranking **M**AX **Sf L**T by de-linking the lexical tone of the suffix in preference for the lexical tone of the stem.

5.3.4.1.2 Type I (b) derivation

 $^{+H}$ nōng+sōr \rightarrow ^Mnōng^{+H}sōr

In this type of derivation, the suffix is not inherently specified with a tone. As the derived word is considered as a prosodic word, the lexical tone of the stem is transferred to the suffix to satisfy **ALIGN-R** (**PRWD**, **LT**). The stem is then specified with a default mid tone. The OT analysis of this type of derivation is shown below:

Max Sf Lt, Specify T, Specify (LT, Prwd), Align-R (prwd, lt), Dep-LT (H, L), Align-L (dt, prwd), No Long LT

>>

INPUT [[nōng]sōr] ^{prwd} H	MAXSF LT	SPECIF YT,	SPECIF Y(LT, PRWD)	ALIGN- R(LT, PRWD)	ALIGN- L(DT, PRWD)	DEP4LT (H,L)	NO LONG LT	*DISASS OCIATE	DEP-DT	NO LONG DT	MAX SIMLT
a. [[nōng]sōr] ^{prwd} H		*!		*!	*!						
b. [[nōng]sōr] ^{prwd} H H				*!	*!	*!					
c. [[nōng]sōr] ^{prwd} DT Low						*!			*		*
☞d. [[nōng] sōr] prwd DT H									*		

*DISASSOCIATE, DEP- DT, NO LONG DT, MAX Stm LT

Table 9

In table 9, the input is a prosodic word and it is not specified with any tone on the right of the prosodic word. Moreover the leftmost syllable of the prosodic word is specified with a lexical tone H (igh).

Hence, *candidate a* violates the high-ranking constraint **SPECIFY T** as the rightmost syllable of the prosodic word is not specified with any tone. It also violates the high-ranking constraint **ALIGN-R** (**PRWD**, **LT**). As there is no lexical tone aligned on the right of the prosodic word, this constraint is violated. Moreover, in the derivation there is no lexical tone associated with the left of the prosodic word. Hence the constraint **ALIGN-L** (**DT**, **PRWD**) is also violated. Considering the violation of the high-ranking constraints, *candidate a* cannot emerge as an optimal candidate.

Candidate b violates ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD) and DEP-LT (H, L). As *candidate b* has two lexical tones in the same prosodic word, it violates ALIGN-R (PRWD, LT). As there is no default mid tone specified on the left of the prosodic word, it also violates the constraint ALIGN-L (DT, PRWD). As a new lexical tone is inserted the high-ranking constraint DEP-LT (H, L) is violated. Moreover, we assume that the universal constraint OCP is underlyingly dominant in Bodo, which prohibits an output like *candidate b*.

Candidate c has a new lexical tone, L(ow) associated with it. Therefore it violates the high-ranking constraint **DEP-LT** (**H**, **L**), which prohibits the insertion of a new lexical tone. As a default tone is inserted in the analysis the candidate also violates the constraint **DEP-DT**. The violation of the high-ranking constraint makes the candidate non-optimal.

Candidate d satisfies all the constraints in the analysis. It violates the low-ranking constraint **D**EP- **D**T. It however does not prevent *candidate d* from becoming the optimal candidate.

5.3.4.1.3 Type I (c) derivation

$pho+^{L}thang \rightarrow ^{M}pho^{L}thang$

In this type of derivations, a tonologically unspecified prefix is attached to a stem. However, the derivation is considered to be a prosodic word. Here the prefix is associated with a default, mid tone, whereas the tonal specification of the stem is preserved.

MAX Sf LT, SPECIFY T, SPECIFY (LT, PRWD), ALIGN-R (PRWD, LT), DEP-LT (H, L), ALIGN-L (DT, PRWD), NO LONG LT

>>

	*Disassociate, Dep- dt, No Long DT, Max Stm Lt											
INPUT [[pho]thang] ^{prwd} L	MAXSF LT	SPECIF YT,	SPECIF Y(LT, PRWD)	ALICN- R(LT, PRWD)	ALIGN- L(DT, PRWD)	DEP-LT (H,L)	NO LONG LT	*DISASS OCIATE	DEP-DT	NO LONG DT	MAX SIMLT	
a. [[pho]thang] ^{prwd} L(ow)		*!			*!							
b. [[pho]thang] ^{prwd} LT L(ow)				*!	*i	*i						
c. [[pho]thang] ^{prwd} DT L(ow)									*			

Table 10

In the analysis above, *candidate a* is associated with no tone on the leftmost syllable. Hence, it violates the high-ranking constraint **SPECIFY T**. It also violates the high-ranking constraint ALIGN-L (DT, PRWD) as there is no default, mid tone specified with the left of the prosodic word. Therefore *candidate a* cannot be optimal.

Candidate b has a lexical tone inserted on the left of the prosodic word. It is a violation of the high-ranking constraint **DEP-LT (H, L)** as this constraint prohibits the insertion of new lexical tones. The lexical tone insertion also violates the high-ranking constraint **ALIGN-R** (**PRWD**, LT). As there is no default, mid tone specified with the left of the prosodic word, a serious violation of the high-ranking constraint **ALIGN-L** (**DT**, **PRWD**) is prompted. Hence, *candidate b* cannot be the optimal candidate.

Candidate c fulfills all the high-ranking constraints in the analysis. Although it violates the low-ranking constraint **D**EP- **D**T, it is not important in the analysis. Hence, *candidate c* emerges as the optimal output in the analysis.

Hence, we saw in section 5.2.1 that the OT analysis of the derived polysyllables is possible with the existing constraints used for the OT analysis of non-derived polysyllables with the addition of the high ranking MAX Sf LT and the down ranked MAX Stm LT.

5.3.4.2 The –mon suffix¹

In our previous discussion we assumed that the $-m\bar{o}n$ suffix is not specified with an inherent tone. The lexical tone of the stem is spread to the suffix as seen in figure 5.3 and 5.4. Another interesting observation in this regard is the fact that this kind of alteration is restricted only to the stems with a high tone.

¹ It is observed that that the $-m\bar{o}n$ suffix is attached only to stems specified with high tones. Apparently we do not have any explanation for that. We assume that to be a lexical gap in Bodo grammar. It is possible to explain the gap in terms of constraint hierarchy. We colclude that the constraint **NO LONG LOW T** dominates the constraint **NO LONG HIGH T.** However, if would be interesting to test what native speakers do when the $-m\bar{o}n$ suffix is attached to a stem with a low tone.



In figure 5.3 the non-derived lexical item /laisri/ is seen to be specified with a high tone. In figure 5.4 the –mon suffix is attached to the non-derived lexical item. It is observed that the high tone of the rightmost syllable of the non-derived lexical item is now spread to the suffix –mon.

We explain the case of $-m\bar{o}n$ suffixation with the help of a morphological explanation. We regard the derived lexical item as a single prosodic unit. At the same time the stem of the lexical item also preserves its identity as a distinct prosodic word. The morphological structure of the derived lexical item in case of $-m\bar{o}n$ suffixation is shown below:

In Bodo **SPECIFY** (**LT**, **PRWD**) is a high-ranking constraint. Hence, $-m\bar{o}n$, which is regarded as a prosodic word and which is not inherently specified with a lexical tone, needs a lexical tone specification. Hence the following situation arises:

2.



In 2, the lexical tone of the prosodic word (=the stem) spread to the right edge of the derived prosodic word. It satisfies the need of the derived prosodic word to have a lexical tone. In the derived words there are two right prosodic word boundaries, both of which have to be satisfied. Thus we see that no new constraints are needed to explain the tonal pattern of the derived forms.

However the constraint **No Long LT** moves lower in the constraint hierarchy. This ranking, therefore, allows lexical tone spread in $-m\bar{o}n$ type of derivations as shown below:



The illustration shows that initially the suffix -mon is not specified with any tone. However in the derivation, the lexical tone (High, in -mon type of suffixation) of the stem spreads to the suffix. Hence, it leaves no TBU unspecified with a tone.

The constraint hierarchy of the constraint necessary for –mon type of derivation looks as below:

MAX Sf LT, SPECIFY T, SPECIFY (LT, PRWD), DEP-LT (H, L), ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD)>>*DISASSOCIATE, DEP-DT, NO LONG DT, NO LONG LT, MAX Stm LT

Table 11 shows how these constraints select an optimal candidate:

INPUT σσ]σ DT LT	MAX SF LT	SPECF Y T	SPECF Y (LT, PRWD)	ALIG N-R (PRW D, LT),	DEP- LT (H, L)	ALIG N-L (DT, PRWD)	LOCA L	*DISA SSOCI ATE	DEP- DT	NO LONG DT	NO LONG LT	MAX STM LT
a.σ σ]σ DT LT		*!		!*								
b.σ σ]σ DT LT DT				!*		*!			*			
c.σ σ] σ DT LT LT i j					*!							
d.σ σ] σ ^(*) / DT LT											*	

Table 11²

In the analysis above, *candidate a*, the suffix is not specified with a tone. Hence, it violates the constraint **SPECIFY T.** As the right edge of the prosodic word is not specified with a tone, *candidate a* also violates the constraint **ALIGN-R** (**PRWD**, **LT**). The constraint **ALIGN-R** (**PRWD**, **LT**) is a high-ranking constraint. Hence, its violation is serious and the candidate cannot be an optimal output.

Candidate b has the right edge of the prosodic word specified with a default, mid tone. Hence, it violates the constraint **ALIGN-L** (**DT**, **PRWD**). As **ALIGN-L** (**DT**, **PRWD**) is a high-ranked constraint, its violation prevents *candidate b* from becoming the optimal candidate. As the right edge of the prosodic word is not specified with a lexical tone, the candidate also violates the high-ranking constraint **ALIGN-R** (**PRWD**, **LT**).

Candidate c is specified with two non-identical lexical tones on the right edge of the stem and on the suffix. This is a **Dep-LT** (**H**, **L**) violation as a lexical tone is inserted in this

² There is a possibility of a candidate like $\sigma \sigma]\sigma$

However, this candidate violates the constraint **DEP-LT** (**H**, **L**), as a new lexical tone is inserted into the derivation. Moreover it also violates the underlyingly dominant constraint **OCP** as two adjacent identical items are prohibited by **OCP**.

derivation, whereas the high-ranking constraint **DEP-LT** (**H**, **L**) prohibits the insertion of any lexical tone.

Candidate d fulfills all the constraints except the dominated constraint No Long LT. *Candidate d* is specified on the right edge with a spread lexical tone, violating the constraint No Long LT. However, its violation is not serious as No Long LT is a dominated constraint in the analysis. As *candidate d* does not violate any of the high-ranked constraints, it emerges as the optimal candidate.

With the help of the constraint ranking suggested above, the following case of tone alteration in Bodo can be taken care of:

^Mlay^{+H}sri + mon
$$\longrightarrow$$
 ^Mlay^{+H}sri^{+H}mon 'Laysri and others'.

Table 12 shows how the optimal candidate ^Mlay^{+H}sri^{+H}mōn is chosen from a variety of possible candidates.

INPUT lay sri] mōn DT LT	MAX SF LT	SPEC FY T	SPEC FY (LT, PRW D)	ALIG N-R (PRW D, LT),	DEP- LT (H, L)	ALIG N-L (DT, PRW D)	LOC AL	*DIS ASSO CIAT E	DEP- DT	NO LON G DT	NO LON G LT	MAX STM LT
a.lay sri] mōn DT LT		*!		!*								
b.lay sri] mōn DT LT DT				!*		*!			*			
c.lay sri] mōn DT LT LT i j					*!							
d.lay sri] mōn DT LT LT					*!							
e.lay sri] mōn @ J DT LT											*	

MAX Sf LT, SPECIFY T, SPECIFY (LT, PRWD), DEP-LT (H, L), ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD)>>*DISASSOCIATE, DEP- DT, NO LONG DT, NO LONG LT, MAX Stm LT

Table 12

The analysis of table 11 is also valid for table 12.

5.3.4.3 The *-ho* suffix

-ho is a causative suffix in Bodo. In the discussion on -ho suffixation in the previous chapter, we came to the conclusion that the suffix -ho is inherently specified with a tone. When the suffix is attached to a stem the inherent specifications of both the stem and the suffix are preserved. Hence a case like the following is noticed in the Bodo causativization:

^M**pho**^{+H}**thai** '*believe*' + ^L**ho** \longrightarrow ^M**pho**^{+H}**thai**^L**ho** '*to make believe*' Hence it is clear that in -ho type of derivation the inherent tonal specification of the stem is preserved in the output. Even the inherent tonal specification of the suffix is preserved in the output. Unlike the $-m\bar{o}n$ suffix, the -ho suffix can be attached to a stem with both high and low tonal specification.



If we look at the picture 5.5 of the suffix -ho, we see that it is inherently specified with a low tone.

These facts about the -ho suffix prompts us to assume that in this type of suffixation the stem and the suffix behave as two distinct prosodic words. The morphological representation can be shown as following:

As the stem and the suffix behave as two distinct prosodic units, they follow the tone assignment system of the prosodic words, discussed in section 3.3.1 of Chapter III of this thesis. Hence, the constraint ranking and analysis proposed for $-m\bar{o}n$ type of affixation is valid for -ho type of affixation also.

Hence, the following ranking of constraints is suggested for *-ho* type of suffixations:

MAX Sf LT, SPECIFY T, SPECIFY (LT, PRWD), DEP-LT (H, L), ALIGN-R (PRWD, LT), ALIGN-L (DT, PRWD)>>*DISASSOCIATE, DEP- DT, NO LONG DT, NO LONG LT, MAX Stm LT

INPUT [[pho thai] ho] DT LT LT	MAX SF LT	SPECIF Y T	SPECIF Y (LT, PRWD)	ALIGN- R (LT, PRWD)	ALIGN- L (DT, PRWD)	DEP-LT (H, L)	*DISAS SOCIA TE	DEP- DT	NO LONG LT	NO LONG DT	MAX STM LT
a.[[pho thai] ho] ↓ DT LT	*!						*		*		
b.[[pho that] ho]			*!	*!			*			*	*
c.[[pho thai] ho] ☞ │ │ │ DT LT LT											

Table 13

In Table 13 the OT analysis of a -ho type of suffixation is shown. In the input the stem /phothai/ is a prosodic word. Hence, it has a default mid tone associated with its left edge and a lexical tone on its right edge. Similarly, as -ho also behaves as a prosodic word, it has a lexical tone associated with it.

In the analysis, in *candidate a* the inherent lexical tone of the suffix is lost and the lexical tone of the stem spreads to the suffix. It violates the high-ranking **Max Sf Lt** and also the low-ranking constraint **No Long LT**. This constraint prevents a lexical tone from being associated with more than one syllable. As the lexical tone in *a* is associated with two TBUs, it is a violation of the constraint **No Long LT**. As the inherent tone of the suffix - *ho* is disassociated, the constraint ***DISASSOCIATE** is violated. However, as ***DISASSOCIATE** is a dominated constraint, its violation does not matter to the analysis.

In *candidate b* the lexical tone of the stem is deleted and the default tone is spread to the rightmost syllable of the stem. As the stem is considered to be a prosodic word, the deletion of the lexical tone is a serious violation of the constraint SPECIFY (LT, PRWD) SPECIFY (LT, PRWD) which says that every prosodic word should be associated with a lexical tone. This candidate also violates the high-ranking constraint ALIGN-R (PRWD, LT) and the low ranking constraint MAX Stm LT. As the inherent lexical tone of the stem is deleted, it also prompts the violation of the low ranking constraint *DISASSOCIATE. Considering the serious constraint violations *candidate b* cannot be the optimal output.

Candidate c fulfills all the high-ranking constraints in the analysis. Hence, it emerges as the optimal output.

The -ho affixes we have examined permit us to come to the conclusion that derived forms in Bodo do not require any major re-ranking or introduction of new constraints (only three) in the constraint hierarchy suggested for non-derived words. However, a word of caution: as we said earlier (Chapter IV), more work needs to be done in the field of morphological derivations in Bodo before we can arrive at a definitive statement regarding tone and morphological derivations in the language.

5.4 Conclusion

In this chapter, for the OT analysis of the morphophonemics of Bodo we have used two different types of alignment constraints. They are ALIGN-R (PRWD, LT) and ALIGN-L (DT, PRWD). ALIGN-L (DT, PRWD) says that each default tone should align with the left edge of the domain. Whereas ALIGN-R (PRWD, LT) says that the left edge of the domain should be specified with a lexical tone. Earlier we tried using the ALIGN-R (LT, PRWD) constraint for lexical tones. But later we found out that in case of derived polysyllables the lexical tone may be assigned to two syllables on the right of the prosodic word. Hence we had to modify the constraint to ALIGN-R (PRWD, LT) for the benefit of the analysis of the derived polysyllables.

Given morphological assumptions, the same constraint hierarchy that was argued for nonderived words can be extended to morphologically derived forms. But more extensive work needs to be done before broad generalization regarding tone and morphology in Bodo.

Appendix I

List of derived and non-derived lexical items used in this work

 \mathbf{A} $\mathbf{a}^{L}d\bar{\mathbf{o}}\mathbf{y}$ $\mathbf{a}^{H}d\bar{\mathbf{o}}\mathbf{y}$ the calf of the leg $\mathbf{a}^{H}\mathbf{khay}$ hand $\mathbf{a}^{L}\mathbf{khay}$ fried paddy $\mathbf{a}^{H}\mathbf{may}$ may maternal uncle $\mathbf{a}^{L}\mathbf{may}$ be vexed $\mathbf{a}^{L}\mathbf{pha}$ my father $\mathbf{a}^{H}\mathbf{pha}$ palm of the feet

B

^Hbar v. jump ^Lbar *n*. day ^Lbar *n*. wind Lbar v.bloom ^H**bar^Lho** to make one jump ^H**baw** v. dedicated to god ^Lbaw v. to forget ^Hbay v. break ^Lbay v. buy **be^Lkheng** make open **be**^L**khew** make loose, make open ^H**beng** *v*. watch ^Lbeng v. agree ^H**bi** v.beg ask pray ^Lbi pron. he or she **bi**^L**phōr** these **bi^Hsōr** they bi^Lsi wife bisi^Hni their **bisi^Lni** tear off **bi^Hthang** he (honorific) **bi^Hthang^Hmon** they (honorific) **be^Hma** spider **be^Lma** thick, bulky **bi^Ldōy** egg of birds **bi^Hdōy** his uncle **bo^Hkha** make separate ^H**bu** *v*.strike, beat ^Lbu v. swell

bu^Hkhu make fall down, make uproot ^H**bung** *v*. speak ^L**bung** *v*. fulfil

D

da^Hbōr moment da^Lbōr tub da^Llang bridge da^Hlang do not take ^Hdang v. touch ^Ldang v. clean or clear ^Hdaw bird daw^Hphōr birds ^Hded big ^H $d\bar{o}$ *n*. cooking pot ^L $d\bar{o}$ *v*. erect a post $^{\rm H}$ dōy v. lay eggs, place; on a head L **d** $\bar{\mathbf{o}}\mathbf{v}$ *n*. water $^{L}d\bar{o}y$ v. sweeten, feel sweet doy^Hkhar run by holding ones hand doy^Hkhar run by holding on hands ^Hdung deep

E

e^Hlo bring liquid with fingers
e^Llo this only
endiphi^Hphang castor oil plant
^Ler v. grow, increase
^Her v. move by a finger or handle
er^Hso more with a speed
er^Hso hide
^Lew v. plough
^Hew v. cut the jungle to make it clear
^Hew v. fry
ew^La n. one does not plough
ew^Ha v. one does not fry
(^H**ga** be separate ^H**gab** n. color ^Lgab v.weep ga^Hhay short, low ga^Lhay main, principal, head gahe^Hnai get a knock on a leg galang^Ho goes on treading $\begin{array}{c} \mathbf{L}_{gan} \quad v. \text{ wear} \\ \mathbf{L}_{gan} \quad n. \text{ music} \\ \mathbf{H}_{gaw} \quad v. \text{ tear or split} \end{array}$ L **gaw** v. shoot by arrow or gun shell ge^Hble become powder ge^Hble melt into **ge^Lbeng** straight ge^Hbeng thin ge^Hbla *adj*. Turn into powder **ge bla** *uaj*. Full into pov **ge^Hbla** whenever ^L**geng** open automatically ^L**gew** open ^H**gi** be afraid of Hgi^Lho make afraid Hglob v. devour Lglob v. fully completely go^Hba thin go^Lba embrace go^Lba vomit go^Lmo yellow **go^Hmo** be surprised ^Lgong *adj*. bent ^Lgong *v*. bend ^Hgong *n*. horn **go^Hsong** stand up **go^Hong^Hho** make one stand up go^Hthang green go^Lthang living, alive go^Ltho child gotho^Lphōr children go^Hza red **go^Lza** empty **gu** put off, up root **gu^Hbung** other, thick gu^Lbung be muddy gu^Lbung with good sounds gu^Ldung hot gu^Hdung hole

gunthu^Hthri the forepart of the nose above the chin

Η

^Hha land ^Lha be able ^Lha cut by pressing hab^La does not enter **ha^Lba** marriage task, work **hadong**^H**mon** cut by pressing ^H**ham** v. be lean and thin ^Lham *v*. come around, be recovered **ha^Hsu** measure land ha^Lsu pass urine ^Lhay be lower hinzawsi^Hkhla matured girl ^Hhor send ^Lhor night ^Lhor bend, hang ho^Hso drive ho^Lso pour ^Hhu land ^Lhu rub

K

^Lkha bind, tie ^Hkha pluck kha^Llay do perform kha^Hlay refused things, surplus ^Hkham boiled rice ^Lkham drum, burn ^Lkham^Lho to make burn khamgli^Hdong has been burnt khamho^Hbay has smelt after burning kha^Hnay hair kha^Lnay to bind ^Hkhar run ^Lkhar smell like a fish ^Hkheb pinch ^Lkheb trip khe^Lray national festival of the Bodos khe^Hray mustard seed cake kherkha^Hni elbow khigra^Hno latrine ^Hkhon arrange clothes ^Lkhon sing occasion ^Lkhon pick up one by one ^Hkhu make naked, remove skin ^Lkhu throw ^Hkhur dig soil by nail ^Lkhur cause to itch khu^Hser sugar cane khu^Lser prick with nails khuseng^Hgra cicada (an insect) khuser^Haw in the sugarcane

L

^Hlaw be long ^Llaw gourd lay^Hsri a proper name lay^Hsri^Hmon laysri and others ^Hlir write, plaster wall ^Llir be heavy ^Hlom be merged in water ^Llom be ill lu^Hkhang finish work lu^Hkhang finish pouring

Μ

man^Hsi man man^Hsi^Hphōr men ^Lmaw move ^Lme^Hthay song, music ^Lme^Lthay molasses mi^Hthi know, understand mi^Hthi^Lho to make know ^Lmō^Ldōm body ^Lmō^Ldōm sweet melling ^Hmōn get ^Lmōn ripen, boil mōnsada^Hza be not sorry ^Lmo^Hsow cow

N

naphithi^Hkhri a kind of small fish (Assamese: Puthi)
^Hno house
^Lno offer
^Hnōng you (singular)
nōng^Hsōr you (plural)
nōng^Hthang you (singular, honorific)
nōng^Hthang^Hmon (plural, honorific)
no^Hphōr houses

0

^Hon love ^Lon rice powder ^Lon spread a loom

P

pha^Lham make cured **pha^Lhay** make lower **phan** v. twist ^Lphan v. sell **^Hphe** *v*. be intoxicated ^L**phe** v. be mild **phe^Hded** to make big **phe^Hreb** make weak **phe^Lsem** make less **phiding**^H**nay** to turn around **phi^Llir** make heavy phi^Hsa small ones **phi^Hsa** son **phisi^Lri** make fall down **pho^Lthaw** make tasteful **pho**^L**thang** give life, save **pho**^H**thay** believe **pho^Hthai^Lho** make believe ^H**phōy** n. plant seeds ^L**phōy** v. come **phu^Hdung** make deep ^H**phung** *n*. morning ^L**phung** *v*. be fast and fleshy **phu^Lrung** make soft

R

^Hran become dry ^Lran divide ^Lraw voice raw phōr voices ^Hray v. harm by magical power ^Lray v. abuse ^Lray v. speak ^Hreb be mild ^Hro v. leak out ^Lro v. stop going rongza^Hdong has become happy ^Lrung become mild

S

^Lsa v. place a fishing instrument net etc. \mathbf{H} sa *n*. north *v*. make rope ^Lsa *v*. ache ^Hsan v. count , think ^Lsan *n*. sun ^Lsem decrease ^Hser v. examine by pressing ^Lser *v*. leak water ^Lser *n*. a see ^Lsew *n*. musical tune ^Hsew *v*. be rotten ^Lsew *n*. motion ^Hsi cloth ^Lsi wet si^Hgi make afraid si^Hling be separate si^Lling make pieces si^Lri fall down $s\bar{o}^{L}gab$ make weep $s\bar{o}^{L}maw$ make moving so^Lmay be fit, decorate so^Hmay oath ^Hsong cook ^Lsong seek opportunity, find out. ^Hsōr all around ^Lsōr who ^Lsōr iron $^{\rm H}$ su wash, saw, thorn ^Lsu weigh, measure

su^Lbung make speak
su^Hgu nine
sugu^Lbung make muddy

Τ

^Hthab v. paste ^Lthab adv. At once thabay^Lbay wander thabay^Hdong has walked ^Hthang v. go ^Lthang v. live thangphla^Hdong has pretended to go ^Lthaw n. oil ^Lthaw v. strike with force thi^La steep thi^La a parrot like bird thi^Ha does not keep for one ^Lthōn v. urge on, incite ^Lthōn v. tell

U

^Hun sharpen ^Lun behind

Ζ

^Hza eat
^Lza be, become
za^Hdong^Hmon ate
zapha^Hbay has already eaten together
za^Hslay eat or take by barter system
za^Lslay be or exist by barter system
zay^Hō eats
zay^Lō is, exists
zen^Lnay to begin
zen^Hnay to be defeated
^Lzi v.tear
^Hzi cloth
zi^Lbow serpent
zibow^Lphōr serpents

Appendix II

Selected Pitch Pictures of the Analysed Bodo Words

Monosyllables*



*For gloss, refer to Appendix I



























































































125















о

о

0

0.398141

0.458367

0.473923

129
























0.327029

0.348481

0.337732



































































da^Hbor

da^Lbor











Time (s)

0.554739

Ó





 175_{160} 140_{120} 100_{100} $100_$























ho^Hso

ho^Lso
































Time (s)

0.497098

lu^Hkhang









za^Hlay









Pictures of the 0 tone claimed by Bhattacharya (1977)* In this set of pictures, in the gloss, the tone is marked as in Bhattacharya (1977). For our version of tone assignment for these words, please refer to Appendix I.



¹za⁰dong⁰mon





²en¹di⁰phi¹phang

²hin³zaw⁰si¹khla



²mon²sa⁰da³za



Plural Suffixation

A. The null suffix



mo^Hsow

 $mo^{H}sow - \emptyset \rightarrow mo^{H}sow$







B. -mon suffix





Time (s)

 $nong^{H}thang-mon \rightarrow nong^{H}thang^{H}mon$



















^Hno-phor→no^Hphor







zi^Lbow -phor→zibow^Lphor













phe^Hded

















phu^Lrung





D. pho-

^Lthaw

pho^Lthaw







pho^Hthang



E. pha-



pha^Lham









be^Hkheng























^Hbar^Lho







pho^Hthai^Lho





Appendix III

List of OT constraints used in this work

ALIGN-L (DT, PRWD): Each default tone should align with the left edge of the domain.

ALIGN-R (PRWD, LT): Each right edge of the domain should align with the T.

DEP-DT: No insertion of default tones.

DEP-LT (H, L): No insertion of High and Low lexical tones.

*DISASSOCIATE: No removal of association lines.

MAX Sf LT: No deletion of suffix lexical tones.

MAX Stm LT: No deletion of stem lexical tones.

No Long DT: A default tone must be associated with at most one TBU.

No Long LT: A lexical tone must be associated with at most one TBU.

NoContour: A TBU may be associated with at most one tone.

OCP: Adjacent identical elements are prohibited.

SPECIFY (LT, PRWD): A prosodic word must be associated with a tone.

SPECIFY T: A TBU must be associated with a tone.
Bibliography

Archangeli, Diana and D. Terence Langendon (ed.s).1997. *Optimality Theory an Overview*. Blackwell Publishers, UK.

Baro, Madhu Ram. 1991. The Bodo Language. Guwahati.

Bhattacharya, Pramod Chandra. 1977. *A Descriptive Analysis of the Boro Language*. Department of Publication, Guwahati University.

Burling, Robbins.1959. "Proto-Bodo". Language 35:433-53

Choudhury, Bhupendra.1958. Some Cultural and Linguistic Aspects of the Garos. Lawyers' Book Stall, Guwahati, Assam.

Fanai, T. Lalrindiki.1992. Some Aspects of the Lexical Phonology of Mizo and English: An Autosegmental Approach. Ph.D. Thesis, CIEFL, Hyderabad, 500007

Gowda, KS Gurubasava.1975. Ao Grammar. CIIL, Mysore, 570006.

Halvorsrud, H.1959. *Boro Grammar*. Rev. H.Halvorsrud for The Boro Literature Board, P.O. Haltugaon, Goalpara, Assam.

Jeyapalli, V.Y. 1987. Karbi Grammar. CIIL, Mysore, 570006.

Joseph, U.V. and Robbins Burling.2001. "Tone Correspondence Among the Bodo Languages". Linguistics of the Tibeto-Burman Area, 24.2:41-55.

Kager, Rene.1999. Optimality Theory. Cambridge University Press.

Kiparsky, Paul. 1973. "*Elsewhere in Phonology*". A Festschrift for Morris Halle.Holt, Rinehart and Winston, Inc.

Lalrindikii.1989. Some Aspects of Autosegmental Phonology of English and Mizo. M.Litt. Dissertation, CIEFL, Hyderabad, 500007

McCarthy, John J.2002. *A Thematic Guide to Optimality Theory*. Cambridge University Press.

Ravindran, N.1974. Angami Phonetic Reader. CIIL, Mysore, 570006.

Singh, Inder.1975. Manipuri Phonetic Reader. CIIL, Mysore, 570006.

Spencer, Andrew.1991. Morphological Theory.Blackwell Publishers,UK.

Sreedhar, M.V. 1976. Sema Phonetic Reader. CIIL, Mysore, 570006.

Thirumalai, M.S.1972. Thaadou Phonetic Reader. CIIL, Mysore, 570006.

Weidert, Alfons.1987. *Tibeto-Burman Tonology*. John Benjamins Publishing Company, Amsterdam/Philadelphia.

Yip, Moira.2002. Tone.Cambridge University Press.